Biochimica et Biophysica Acta 1827 (2013) 114-135

Contents lists available at SciVerse ScienceDirect



Biochimica et Biophysica Acta



# Diversity and evolution of bioenergetic systems involved in microbial nitrogen compound transformations $\overset{\triangleleft}{\succ}$

### Jörg Simon <sup>a,\*</sup>, Martin G. Klotz <sup>b,\*\*</sup>

<sup>a</sup> Microbial Energy Conversion and Biotechnology, Department of Biology, Technische Universität Darmstadt, Schnittspahnstr. 10, 64287 Darmstadt, Germany <sup>b</sup> Department of Biology, University of North Carolina, 9201 University City Boulevard, Charlotte, NC 28223, USA

#### ARTICLE INFO

Article history: Received 17 May 2012 Received in revised form 10 July 2012 Accepted 19 July 2012 Available online 25 July 2012

Keywords: Anammox Biogeochemical nitrogen cycle Denitrification Nitrification Quinone/quinol-reactive protein module Respiratory nitrate and nitrite ammonification

#### ABSTRACT

Nitrogen is an essential element of life that needs to be assimilated in its most reduced form, ammonium. On the other hand, nitrogen exists in a multitude of oxidation states and, consequently, nitrogen compounds (NCs) serve as electron donor and/or acceptors in many catabolic pathways including various forms of microbial respiration that contribute to the global biogeochemical nitrogen cycle. Some of these NCs are also known as reactive nitrogen species able to cause nitrosative stress because of their high redox reactivity. The best understood processes of the nitrogen cycle are denitrification and ammonification (both beginning with nitrate reduction to nitrite), nitrification (aerobic oxidation of ammonium and nitrite) and anaerobic ammonium oxidation (anammox). This review presents examples of the diverse architecture, either elucidated or anticipated, and the high degree of modularity of the corresponding respiratory electron transport processes found in Bacteria and Archaea, and relates these to their respective bioenergetic mechanisms of proton motive force generation. In contrast to the multiplicity of enzymes that catalyze NC transformations, the number of proteins or protein modules involved in connecting electron transport to and from these enzymes with the quinone/ quinol pool is comparatively small. These quinone/quinol-reactive protein modules consist of cytochromes b and c and iron-sulfur proteins. Conclusions are drawn towards the evolutionary relationships of bioenergetic systems involved in NC transformation and deduced aspects of the evolution of the biogeochemical nitrogen cycle are presented. This article is part of a Special Issue entitled: The evolutionary aspects of bioenergetic systems.

© 2012 Elsevier B.V. All rights reserved.

#### 1. Introduction

The redox transformation of nitrogen compounds (NCs) is the basis of nature's biogeochemical nitrogen cycle (Fig. 1). The involved reactions and pathways are manifold and serve in nitrogen assimilation

and nitrogen-based catabolism including respiration as well as detoxification of harmful NCs ("reactive nitrogen species"; see Table 1 for some general properties of the biologically most relevant NCs) [1–3]. In particular, the participation of a diverse range of microorganisms from the bacterial and archaeal domains is crucial for the maintenance of the nitrogen cycle. Important reactions of the nitrogen cycle include (i) anaerobic reduction of nitrate  $(NO_3^-)$  to nitrite  $(NO_2^-)$ , (ii) reduction of nitrite to produce nitric oxide (NO) and nitrous oxide (N<sub>2</sub>O) (during both anaerobic "classic" denitrification and aerobic "nitrifier" denitrification) and dinitrogen (during classic denitrification): (iii) reduction of nitrite to ammonium (ammonification pathways), (iv) aerobic ammonium oxidation to yield hydroxylamine, nitrite and eventually nitrate (nitrification), (v) dinitrogen (N<sub>2</sub>) fixation, (vi) anaerobic ammonium oxidation (anammox) to produce N<sub>2</sub> and (vii) a recently described reaction of nitric oxide dismutation resulting in N<sub>2</sub> and O<sub>2</sub> production (Fig. 1). The processes (i), (ii), (iii), (iv), (vi) and (vii) belong to different modes of respiratory energy metabolism that drive ATP generation in the corresponding cells. In general, examined organisms that live under anoxic conditions are capable to catalyze only one of the denitrification, ammonification, anammox or NO dismutation processes.

Respiratory NC transformation is catalyzed by a diverse range of metalloproteins that are organized in electron transport chains

Abbreviations: Amo, ammonium monooxygenase; *Ca., Candidatus*; Cu-MMO, copperdependent membrane monooxygenase family; DMSO, dimethyl sulfoxide; ETC, electron transport chain; Fe/S, iron-sulfur center; *GOE, Great Oxygenation Event*; Hao, hydroxylamine oxidoreductase; HCO, heme-copper oxidase family; Hzo, hydrazine oxidoreductase; Hzs, hydrazine synthase; Hyd, hydrogenase; Fdh, formate dehydrogenase; MCC, multiheme cytochrome *c* family; Mo-*bis*-MCD, molybdenum *bis* molybdopterin guanine dinucleotide; MK/MKH<sub>2</sub>, menaquinone/menaquinol; N, nitrogen; Nap, periplasmic nitrate reductase; Nar, membrane-bound nitrate reductase; NC, nitrogen compound; NirK, copper nitrite reductase; NirS, cytochrome *cd*<sub>1</sub> nitrite reductase; Nod, nitric oxide dismutase; Nor, nitric oxide reductase; pMmo, particulate methane monooxygenase; *pmf*, proton motive force; Q/QH<sub>2</sub>, quinone/quinol (unspecified); QRP, quinone/quinol-reactive protein; rET, reverse electron transport; Tat, twin arginine translocation; TMAO, trimethylamine *N*-oxide; UQ/UQH<sub>2</sub>, ubiquinone/ubiquinol

 $<sup>\</sup>stackrel{\mbox{\tiny this}}{\rightarrow}$  This article is part of a Special Issue entitled: The evolutionary aspects of bioenergetic systems.

<sup>\*</sup> Corresponding author. Tel.: +49 6151 165203; fax: +49 6151 162956.

<sup>\*\*</sup> Corresponding author. Tel.: +1 704 687 8686; fax: +1 704 687 3128. E-mail addresses: simon@bio.tu-darmstadt.de (J. Simon), mklotz@uncc.edu

<sup>(</sup>M.G. Klotz).

<sup>0005-2728/\$ -</sup> see front matter © 2012 Elsevier B.V. All rights reserved. http://dx.doi.org/10.1016/j.bbabio.2012.07.005



**Fig. 1.** Conversion of NCs serving as substrates in nitrogen cycle processes relevant to this article. Anaerobic respiratory pathways and aerobic nitrification are indicated by grey and white arrows, respectively. The oxidation state of nitrogen atoms is indicated above the NC. Numbers in black decagons refer to the following processes: 1, respiratory nitrate reduction to nitrite; 2, denitrification of nitrite to N<sub>2</sub>; 3, Nrf-dependent ammonification; 4, anaerobic ammonium oxidation (anammox, i.e. comproportionation of nitrite and ammonium to form dinitrogen); 5, Nod-dependent nitrite reduction to dinitrogen; 6, ammonium oxidation to nitrite; 7, nitrite oxidation to nitrate. These metabolic pathways are catalyzed by distinct respiratory enzyme systems that are designated by the NC-converting enzyme: Nar, membrane-bound nitrate reductase; Nap, periplasmic nitrate reductase; Nirš, cytochrome  $cd_1$  nitrite reductase; NirK, copper nitrite reductase; Nor, nitric oxide reductase; Hzo, hydrazine oxidoreductase; Nod, nitric oxide dismutase; Amo, ammonium monooxygenase; Hao, hydroxylamine oxidoreductase; Nxr, nitrite oxidoreductase.

(ETCs) and networks (resulting from interconnected ETCs). These proteins and protein complexes are either bound to membranes, for example the cytoplasmic ("inner") or the anammoxosome membrane, or they are soluble proteins that often reside in compartments outside the cytoplasm such as the periplasm or the lumen of intracellular membrane stacks of Gram-negative bacteria or inside the anammoxosome of the *Brocadiaceae*, the anammox bacteria. In microbes capable of using catabolic transformations of NCs to support growth, electron transport to and from NCs is designed to generate a *proton motive force* (*pmf*), i.e. a transmembrane electrochemical proton potential that drives ATP synthesis via the membrane-bound ATP synthase. In addition, there are cohorts of microbes capable of NC transformations that contribute to known N-cycle processes including N-respiration but cannot support their growth; for instance many methane-oxidizing bacteria can facilitate processes (i) to (v) and (vii) although they do not grow chemolithotrophically using nitrogen compounds [4].

A key redox mediator of the underlying electron transport chains is the membranous quinone/quinol pool that consists of one or more benzo- or naphthoquinones, usually ubi- and/or menaquinones, depending on the organism and its environmental conditions. Quinones are small, freely diffusible lipophilic molecules that are able to take up two electrons and two protons while converted from the oxidized quinone to the reduced quinol state. Various quinone/quinol-reactive proteins (QRPs) are known to contribute to the highly diverse ETCs that catalyze NC conversion. QRPs are divided into donor:quinone dehydrogenases and quinol:acceptor reductases as they catalyze redox reactions situated at either the oxidative or the reductive side during electron transport from a reduced electron donor substrate (the reductant) to an oxidized electron acceptor substrate (the oxidant). Depending on their oxidation state, NCs are functional in both processes (Fig. 1; Table 1).

Many high-resolution structures of NC-converting enzymes and also of a few QRPs have been obtained over the past two decades. On the other hand, the detailed enzymic composition and function of complete ETCs and networks is less well understood. Section 2 of this article introduces some prominent model organisms of respiratory NC turnover and considers the bioenergetic foundations of their catabolic lifestyles. The key enzymes of NC turnover and their connection to one or more distinct QRPs are described in Section 3. Emphasis is placed on ETC modularity and the mode(s) of *pmf* generation in order to separate protonmotive and non-protonmotive steps, which are comparatively addressed in Section 4. Key protein families in respiratory nitrogen metabolism and their assumed evolutionary relationships are presented in Section 5 whereas Section 6 draws conclusions on the evolution of bioenergetic systems involved in NC turnover, mainly on the basis of their modularity. Impacts on the evolution of the extant biogeochemical nitrogen cycle are also discussed.

Note that enzymes having solely assimilatory functions like soluble cytoplasmic nitrate and nitrite reductases as well as dinitrogen-fixing nitrogenases are not discussed in this article as their function is not directly coupled with respiratory metabolism. Furthermore, a detailed structural and functional description of individual enzymes is considered beyond

Table 1

Properties of nitrogen compounds relevant to this article.

Nitrogen compound	Oxidation state	Toxicity to microbial cells	Other properties
Nitrate (NO <sub>3</sub> )	+5	Non-toxic in physiological concentrations.	Anion of strongly oxidizing and toxic nitric acid (HNO <sub>3</sub> ; $pK_a = -1.4$ ).
Nitrogen dioxide (NO <sub>2</sub> )	+4	Toxic orange gas. Arises from the oxidation of nitric oxide by oxygen.	In equilibrium with the colorless gas dinitrogen tetroxide ( $N_2O_4$ ).
Nitrite (NO <sub>2</sub> <sup>-</sup> )	+3	Toxic; binds to cellular iron atoms, for example in hemoglobin.	Forms nitrosonium in acidic conditions according to $HNO_2 + H^+ \rightarrow NO^+ + H_2O$ . Anion of unstable nitrous acid (HNO <sub>2</sub> ; pK <sub>a</sub> =3.4).
Nitric oxide (NO)	+2	Highly reactive toxic radical. Binds to heme iron atoms and Fe/S centers and forms dinitrosyl iron complexes [Fe(NO) <sub>2</sub> ]. Causes nitrosation of thiol groups to form S-nitrosothiols ( $-S-N=0$ ).	Generation of nitrogen dioxide $(NO_2)$ or peroxynitrite $(ONOO^-)$ in the presence of oxygen and superoxide radicals, respectively. Redox-related reactive species are the nitroxyl $(NO^-)$ and nitrosonium $(NO^+)$ ions.
Nitrous oxide (N <sub>2</sub> O)	+1	Chemically inert and non-toxic in physiological concentrations.	-
Nitroxyl (HNO)	+1	Reactive towards nucleophiles, especially thiols.	Dimerizes to hyponitrous acid ( $H_2N_2O_2$ ), which can be dehydrated to $N_2O_2$ .
Dinitrogen (N <sub>2</sub> )	0	Chemically inert and non-toxic.	-
Hydroxylamine (NH <sub>2</sub> OH)	-1	Toxic by binding to heme groups. Irreversible inhibitor of the oxygen-evolving complex in photosystem II.	Mutagenic by uracil-forming cytosine deamination, thus causing $G/C \rightarrow T/A$ transitions in DNA.
Hydrazine (N <sub>2</sub> H <sub>4</sub> )	-2	Highly toxic.	Rocket fuel, forms monohydrate.
Ammonium $(NH_4^+)$	-3	Non-toxic in physiological concentrations.	Cation of ammonia (NH <sub>3</sub> ), which is a toxic uncoupling agent $(pK_a = 9.25)$ .

the scope of this article. Instead, the main focus is set on the modular design and evolutionary relationships of extant bioenergetic systems.

### 2. Metabolic lifestyles of model organisms and consideration of bioenergetic principles

Microbial growth using NCs as respiratory substrates is widespread in heterotrophic and autotrophic members of the domains Bacteria and Archaea. However, compared to the overwhelming amount of data from genomes and environmental metagenomes, only a tiny number of such organisms have been characterized in detail. Taxonomically, most of the well-understood bacteria belong to the phylum Proteobacteria, which is divided into the classes of Alpha-, Beta-, Gamma-, Delta-, Epsilon- and Zetaproteobacteria. Of those, the first three classes form the most recent phylogenetic group that comprises many well-known model bacteria. However, even in these closely related classes, an intriguing variety of respiratory electron transport chain design has been described and the same holds true for the phylogenetically older Delta- and Epsilonproteobacteria. Other bacterial (super) phyla whose study provided significant insight into NC respiration are Nitrospirae, Planctomycetes-Verrucomicrobia, Deinococcus-Thermus, Firmicutes, the NC10 phylum as well as the Cren-, Eury- and Thaumarchaeota. In fact, many recently discovered enzymes and even entire catabolic lifestyles are attributed to microorganisms from one of these less studied phyla, for some of which general microbiological knowledge is rather rudimentary (see Section 3). In the absence of biochemical data, metabolic models are often deduced from genomic information (most importantly by inspecting gene clusters encoding respiratory enzymes and their biogenesis systems), which allows the hypothetical reconstruction of ETCs. It should be kept in mind, however, that this deductive metabolic reconstruction approach might result in unproven ecological predictions, especially in the light of pronounced substrate promiscuity of some NC-converting enzymes.

Microbial respiration is designed for energy conservation, that is to generate ATP by ETC-level phosphorylation, and the organization of the corresponding ETCs generally reflects the energetic constraints of a catalyzed redox reaction. In principle, the free energy change ( $\Delta G$ ) of a given redox reaction is stepwise transduced into an electrochemical potential difference (the *pmf*) across the membrane in order to drive ATP synthesis [5]. This fact represents the unifying theory of bioenergetics introduced as the chemiosmotic hypothesis by Peter Mitchell in the 1960s. The standard free energy change at pH 7.0 ( $\Delta G_0$ ) of a respiratory process depends on standard redox potentials at pH 7.0 ( $E_0$ ) of the electron donor and acceptor substrates (and their respective reaction products) according to the equation  $\Delta G_0 = n \cdot F \cdot \Delta E_0$ , wherein *n* is the number of electrons transferred and F is the Faraday constant (see Table 2 for  $\Delta E_0$  values of selected NC redox pairs). Due to their rather positive E<sub>0</sub> values, the diverse range of NCs is well-suited to serve as terminal electron acceptor of respiratory ETCs, especially when lowpotential substrates like formate, hydrogen gas, diverse sulfur compounds or NADH are used as electron donors in chemoorganotrophic or chemolithotrophic energy-conserving pathways. On the other hand, ammonium and nitrite are poor electron donors, explaining the fact that obligate chemolithotrophic nitrifying organisms usually use oxygen as the sole electron acceptor. In addition, operating a nitrifying electron transport chain by a chemolithoautotroph necessitates reverse electron flow in order to produce NAD(P)H for assimilatory CO2 reduction.

Different kinds of quinone/quinol pairs are characterized by distinct  $E_0^{\circ}$  values. High-potential ubiquinones  $[E_0^{\circ} (UQ/UQH_2) + 110 \text{ mV}]$  are predominantly used by aerobic organisms whereas low-potential menaquinones  $[E_0^{\circ} (MK/MKH_2) - 80 \text{ mV}]$  are better suited for anaerobic respiration when the cellular state is more reduced. From the  $E_0^{\circ}$  values of different NCs presented in Table 2, it becomes clear that quinol oxidation (in particular menaquinol oxidation) in anaerobic respiration using nitrate, nitric oxide, nitrous oxide or hydroxylamine as

terminal electron acceptor is a strongly exergonic reaction. Therefore, in principal, *pmf* generation due to an electrogenic (protonmotive) ETC is energetically possible. Nonetheless, such reactions might be employed as electroneutral (non-protonmotive) steps due to the interchangeable modular architecture of the pertinent ETC components.

The *pmf* has both a chemical ( $\Delta pH$ , dimensionless) and an electrical ( $\Delta \psi$ , dimension mV) component and the formula *pmf* (mV) =  $\Delta \psi - 59 \Delta pH$  describes this correlation in its simplest form.  $\Delta \psi$  is defined as the electrical potential difference between the positively and negatively charged side of the membrane (usually a positive value) whereas the  $\Delta pH$  is defined as the pH difference between both membrane sides (usually negative). Since the *pmf* has a value in the range of about 150 to 200 mV, a redox potential change ( $\Delta E$ ) of at least this magnitude is required during electron transfer from a donor to an acceptor substrate if this step is to be coupled to *pmf* generation. The ratio of protons (transported across the membrane) and electrons (transported from donor to acceptor), i.e. the  $H^+/e^-$  value, corresponds to the quotient of  $\Delta E$  and the *pmf* (at a hypothetical 100%) thermodynamic efficiency). There are different possibilities to generate the *pmf*: (i) proton pumping as exemplified by NADH: ubiquinone oxidoreductase (complex I-type NADH dehydrogenase) or complex IV-type heme-copper oxidases (HCOs), (ii) movement of protons across the membrane via the so-called "Q-cycle" as catalyzed by the cytochrome  $bc_1$  complex (ubiquinol:cytochrome *c* oxidoreductase; complex III) and (iii) the electrogenic redox loop mechanism in which a separation of positive and negative charges across the membrane is achieved by different locations of the active sites for substrates and quinones in respiratory enzymes (or in respiratory electron transport chains) [6]. Classically, a redox loop consists of two quinone/quinol-reactive (multi-subunit) enzymes and a quinone species that serves as redox mediator between them. Bacterial redox loops are commonly found in anaerobic respiratory systems that are driven by  $\Delta G$  values smaller than in chemoorganoheterotrophic aerobic respiration and, consequently, shorter ETCs with a lower number of coupling sites are employed in such cases [6].

## 3. Enzymes and protein modules involved in extant respiratory nitrogen compound transformations

Based on the diversity of NC conversions depicted in Fig. 1, a compilation of the involved respiratory key enzymes is presented in Table 2. Throughout this section, emphasis is laid on the various modes of how such enzymes are connected with the membranous quinone/quinol pool and how NC turnover is envisaged to be coupled to *pmf* generation by the respective ETC. In particular, the taxonomic distribution, modularity and substrate specificity of such ETCs are pointed out.

#### 3.1. Nitrate reductases

Respiratory nitrate reductase complexes belong to either the Nar- or the Nap-type (enzymes 1 and 2 in Table 2). Both nitrate-reducing subunits (NarG or NapA) produce nitrite in an energetically favorable reaction using a quinol as electron donor (Table 2). However, the enzymic composition of the corresponding electron transport chains as well as their contribution to *pmf* generation is highly diverse in bacterial and archaeal species (Figs. 2 and 3) [7-10]. The common feature of the membrane-bound nitrate reductase, the Nar-type, is a heterodimeric NarGH complex that is attached to the membrane by at least one other protein, which is quinol-reactive. The NarGH complex is similar and evolutionarily related to the nitrite-oxidizing NxrAB complexes from anammox bacteria and aerobic nitrifiers (see Sections 3.5, 3.7 and 5.2.2) [11]. The NarG protein binds a molybdenum bis molybdopterin guanine dinucleotide (Mo-bis-MGD) cofactor at the active site of nitrate reduction and a 4Fe-4S cluster while the NarH subunit binds four further iron sulfur clusters (one 3Fe-4S and three

 Table 2

 Selected enzymes and enzyme complexes involved in respiratory nitrogen compound turnover.

Enzyme designation and corresponding standard potential of the relevant redox pair (oxidized/reduced) at pH 7 ( $E_0$ ) <sup>a</sup>	Physiological function	Redox partner	Selected model organism(s) <sup>b</sup>
Niture and interest $NO = +2 = +2U^{+}$ $NO = +U^{-}$	F(NO=NO=) + 0.42 V		
Nitrate reductases: $NO_3^- + 2 e^- + 2H^+ \rightarrow NO_2^- + H_2C$ 1a. nNar/NarGHI (Mo-bis-MGD Fe/S bene b: dimeric)	$(E_0 (NO_3^-/NO_2^-) = +0.43 V)$ Cytoplasmic nitrate reduction	Quinol (oxidized by NarI)	<b>Escherichia coli</b> ( $\gamma$ ), Paracoccus depitrificans ( $\alpha$ )
1b. cNar/NarGHIC (Mo-bis-MGD, Fe/S, heme b, heme c)	Cytoplasmic nitrate reduction	Ubiquinol and periplasmic cytochrome <i>c</i> pool via	Thermus thermophilus (Deinococcus–Thermus)
1c. pNar/NarGH in a complex with other proteins (see text)	Nitrate reduction at the outside of the cytoplasmic membrane	Quinones and possibly small periplasmic electron transfer	Haloferax mediterranei, Haloarcula marismortui (both Euryarchaeota), Pyrobaculum aerophilum
(Mo-bis-MGD, Fe/S) 2a. NapAB (Mo-bis-MGD, Fe/S, heme <i>c</i> )	Periplasmic nitrate reduction	proteins Quinol dehydrogenases NapC and/or NapGH	(Crenarchaeota) <b>Rhodobacter sphaeroides</b> ( $\alpha$ ), Escherichia coli ( $\gamma$ ), Paracoccus denitrificans ( $\alpha$ ), Bradyrhizobium japonicum ( $\alpha$ ), Shewanella oneidensis ( $\gamma$ ), Wolinella succingenes ( $\epsilon$ )
2b. NapA (Mo-bis-MGD, Fe/S)	Periplasmic nitrate reduction	Not known, potentially NapM.	<b>Desulfovibrio desulfuricans</b> $(\delta)$
Nituita nanativa annuna			
Nitrite-reactive enzymes			
(1) NO-genic nitrite reduction: $NO_2 + e^2 + 2H^2$	$\rightarrow$ NO + H <sub>2</sub> O; E <sub>0</sub> (NO <sub>2</sub> /NO) = +0.36 V		<b>— • /</b> .
3. NirS, cytochrome $cd_1$ (hemes $c$ and $d_1$ ; dimeric)	Nitrite reduction	Monoheme cytochromes $c$ (cyt. $c_{550}$ or $c_{551}$ ), NirT or cupredoxin	<b>Paracoccus pantotrophus</b> $(\alpha)$ , <b>Pseudomonas aeruginosa</b> $(\gamma)$ , Pseudomonas stutzeri $(\gamma)$
4a. NirK (Cu <sub>l</sub> and Cu <sub>ll</sub> centers; trimeric)	Nitrite reduction	Monoheme cytochromes $c$ (cyt. $c_{550}$ or $c_{551}$ ) or cupredoxin	Bradyrhizobium japonizum ( $\alpha$ ), Achromobacter cycloclastes ( $\beta$ ), Achromobacter vylosovidans ( $\beta$ )
4b. <i>c</i> NirK (heme <i>c</i> ; Cu <sub>l</sub> and Cu <sub>ll</sub> centers)	Nitrite reduction	Monoheme cytochromes $c$ (cyt. $c_{550}$ or $c_{551}$ ) or cupredoxin	Burkholderia sp. ( $\beta$ ), Bdellovibrio bacteriovorus ( $\delta$ )
4c. CuNirK (three Cu centers)	Nitrite reduction	Monoheme cytochromes $c$ (cyt. $c_{550}$ or $c_{551}$ ) or cupredoxin	Hyphomicrobium denitrificans $(\alpha)$
(ii) Ammonifying nitrite reduction: $NO_2^- + 6e^- + 8H$	$H^+ \rightarrow NH_4^+ + 2 H_2O; E_0(NO_2^-/NH_4^+) = +0.34 V$		
5a. NrfA (5 heme c; one CX <sub>2</sub> CK motif; dimeric)	Periplasmic nitrite ammonification	Pentaheme cytochrome c NrfB	<b>Escherichia coli</b> $(\gamma)$
5b. NrfA (5 heme $c$ ; one CX <sub>2</sub> CK motif)	Periplasmic nitrite ammonification	Tetraheme cytochrome <i>c</i> quinol dehydrogenase CymA	Shewanella oneidensis $(\gamma)$
5c. NrfHA complex (NrfA: 5 heme c; one CX <sub>2</sub> CK motif; NrfH: four heme c; Dimers of NrfHA <sub>2</sub> assembly in crystal structure)	Periplasmic nitrite ammonification	Quinol (oxidized by NrfH)	Wolinella succinogenes $(\epsilon)$ , Desulfovibrio vulgaris $(\delta)$
5d. NrfA (5 heme <i>c</i> ; five CH <sub>2</sub> CH motifs) 6. Onr (8 heme <i>c</i> ; hexameric)	Periplasmic nitrite ammonification Not known. Onr reduces nitrite, hydroxylamiae, culfte, and hydrogen perovide.	Quinol (oxidized by NrfH) Not known	Campylobacter jejuni ( $\epsilon$ ) <b>Thioalkalivibrio nitratireducens</b> ( $\gamma$ )
7. Otr (8 heme <i>c</i> )	Not known. Otr reduces nitrite and hydroxylamine and interconverts tetrathionate and thiosulfate	Not known	Shewanella oneidensis $(\gamma)$
8 eHao (8 heme c)	Most likely nitrite reduction	Cytochrome c of the NanC/NrfH	Campylobacter concisus
0. 61 d0 (0 field <i>c</i> )	wost likely mane reduction	family (?)	Campylobacter concisas, Campylobacter curvus, Campylobacter fetus,
			Nautilia profundicola (all $\varepsilon$ )
(iii) Nitrite oxidoreductase: $NO_2^- + H_2O \rightarrow NO_3^- + 2$ 9a. nNxr/NxrABIC	$e^- + 2H^+$ ; $E_0^{(NO_3^-/NO_2^-)} = +0.43 V$ Cytoplasmic nitrite oxidation	Quinone pool and cytochrome	Nitrobacter hamburgensis ( $\alpha$ )
(Mo-bis-MGD, Fe/S, hemes $b$ and $c$ )		aa3 oxidase	
9b. pNxr/NxrAB (Mo. his MCD. Fo/S. possibly home b)	Periplasmic nitrite oxidation	Quinone pool and cytochrome	Ca. Nitrospira defluvii (Nitrospirae)
9c, aNxr/NxrAB (might form large complex	Nitrite oxidation in anammoxosome	Quinone	Ca. Kuenenia stuttgartiensis
with several other subunits, see text)			(Planctomycetes)
Nitric oxide-reactive enzymes			
(i) N <sub>2</sub> O-genic nitric oxide reductases: $2 \text{ NO} + 2e^{-3}$	$+2H^+ \rightarrow N_2O + H_2O$ ; E <sub>0</sub> (NO/N <sub>2</sub> O) = +118	V	
10a. <i>c</i> Nor/NorBC (hemes <i>b</i> and <i>c</i> , Fe; dimeric)	Nitric oxide reduction	Monoheme cytochromes <i>c</i>	Paracoccus denitrificans ( $\alpha$ ), <b>Pseudomonas aeruginosa</b> ( $\gamma$ )
10b. qNor/NorB (heme <i>b</i> , Fe)	Nitric oxide reduction at the outside of the cytoplasmic membrane	Quinol (MKH <sub>2</sub> )	Pyrobaculum aerophilum (†) Pyrobaculum aerophilum (†) <b>Geobacillus stearothermophilus</b> (Firmicutes), Ralstonia eutropha (β), Neisseria meningitidis (β)
10c. $Cu_A$ -qNor (heme <i>b</i> , Fe, $Cu_A$ )	Nitric oxide reduction at the outside of the cytoplasmic membrane	Menaquinol	Bacillus azotoformans (Firmicutes)
10d. sNor, eNor, gNor	Periplasmic nitric oxide reduction	Monoheme cytochromes c	Diverse Proteobacteria
(ii) Hydrazine synthase (comproportionation of nitr	ic oxide and ammonium): $NO + NH_4^+ + 3e^- + 2H_4^-$	$^{+} \rightarrow N_{2}H_{4} + H_{2}O; E_{0} = +0.06 V$	
11. Hzs (heme <i>c</i> )	Comproportionation of nitric oxide and ammonium in anammoxosome	Cytochromes <i>c</i> in anammoxosome	<i>Ca.</i> Kuenenia stuttgartiensis (Planctomycetes)
(iii) Nitric oxide dismutase (Nod): $2 \text{ NO} \rightarrow N_2 + O_2$ 12. Nod (enzyme yet to be identified)	Nitric oxide reduction in oxygenic	Not known	Ca. Methylomirabilis oxyfera (NC10 phylum)
	acmanication of mane by methanou opins		

(continued on next page)

able 2 (continuea)				
Enzyme designation and corresponding standard potential of the relevant redox pair (oxidized/reduced) at pH 7 $(E_0^{-})^{a}$	Physiological function	Redox partner	Selected model organism(s) <sup>b</sup>	
Nitrous oxide reductase: $N_2O + 2e^- + 2H^+ \rightarrow N_2 + H_2$	$_{2}O; E_{0}(NO/N_{2}O) = +1.35 V$			
13a. NosZ (Cu <sub>A</sub> , Cu <sub>Z</sub> ; Dimer)	Periplasmic nitrous oxide reduction	Monoheme cytochromes c	Pseudomonas stutzeri (y), Paracoccus	
13b. <i>c</i> NosZ (heme <i>c</i> , $Cu_A$ , $Cu_Z$ )	Periplasmic nitrous oxide reduction	(cyt. c <sub>550</sub> or c <sub>551</sub> ), cupredoxin, or NosR/NosXR Monoheme cytochromes c or NosGH quinol dehydrogenase	denitrificans ( $\alpha$ ), Marinobacter hydrocarbonoclasticus ( $\gamma$ ) Wolinella succinogenes ( $\varepsilon$ )	
Ammonia monoovygenase: $NH_{\tau}^{+} + O_{2} + 2e^{-} + H^{+} \rightarrow NH_{\tau}OH + H_{2}O$ : $E_{\tau}^{\prime}$ (NH <sub>2</sub> OH/NH <sub>{\tau}</sub> <sup>+</sup> ) = + 0.74 V				
14. AmoABC (Cu, Fe; trimeric)	Periplasmic ammonia oxidation	Quinone pool (assumed)	Nitrosomonas europaea (β)	
Hydroxylamine oxidoreductase: $NH_2OH + H_2O \rightarrow NO_2$	$\frac{1}{2} + 4 e^{-} + 5H^{+}; E_{0}^{\prime} (NO_{2}^{-}/NH_{2}OH) = +0.10 V$			
15. Hao (8 neme $c$ ; trimeric)	Peripiasmic hydroxylamine oxidation	Cyt. $c_{554}$ (assumed)	Nitrosomonas europaea (B)	
Hydrazine oxidoreductase: $N_2H_4 \!\rightarrow\! N_2 \!+\! 4 \; e^- \!+\! 4 H^+;$ 16. Hzo (8 heme c)	$E_{0}^{\prime }\left( N_{2}H_{4}/N_{2}\right) =-0.75$ V Hydrazine oxidation in anammoxosome	Cytochromes <i>c</i> in anammoxosome	Ca. Kuenenia stuttgartiensis (Planctomycetes)	

<sup>a</sup> Individual enzymes or enzyme complexes are consecutively numbered and their metal/cofactor content as well as their typical multimerization status is provided if reported. <sup>b</sup> Names in bold face indicate that a high-resolution enzyme structure from the respective organism is available. Phyla and proteobacterial classes (Greek letters) are given in parentheses. *Ca., Candidatus.* 

4Fe-4S clusters) [12]. In Proteobacteria (and also in some Grampositive bacteria), the NarGH complex is attached to the cytoplasmic side of the membrane by the quinol-oxidizing diheme cytochrome b membrane anchor Narl that contains five transmembrane helices (Fig. 2A). The two heme groups of NarI are oriented towards opposite sides of the membrane. The NarGHI complex is also called the nNar system since nitrate is reduced at the potential-negative cytoplasmic face of the membrane, the n-side (enzyme 1a in Table 2). Bioenergetically, nNar contributes to *pmf* generation via a redox loop mechanism [6,13]. As depicted in Fig. 2A, quinol oxidation by NarI is catalyzed at the periplasmic side of NarI. Hence, two protons are released into the periplasmic space whereas two electrons are moved across the membrane where they eventually serve in cytoplasmic nitrate reduction coupled to the take-up of two cytoplasmic protons. Overall, quinol oxidation by nitrate using the nNar system results in a transmembrane electrochemical proton gradient with a H<sup>+</sup>/e<sup>-</sup> ratio of one. It is notable that such a system requires nitrate import (and usually also nitrite export), which, in Proteobacteria, is achieved by several distinct transporters (uniport and nitrate/nitrite antiport) [14,15]. Interestingly, bacterial NarG proteins contain a so-called remnant twin arginine translocation (Tat) signal peptide suggesting that primordial bacterial NapGH complexes have been exported and were functional at the periplasmic face of the cytoplasmic membrane [16].

A variation of the nNar system was described in the hyperthermophilic bacterium *Thermus thermophilus* (phylum Deinococcus–Thermus) [17]. In this case, a periplasmic diheme cytochrome c (NarC) was reported to interact with the NarGHI complex that itself seems to form a supercomplex with a membrane-bound NADH dehydrogenase, the Nrc complex (Fig. 2B). The NarGHIC assembly, here designated cNar, presumably catalyzes electrogenic nitrate reduction using ubiquinol as electron donor (enzyme 1b in Table 2). NarC is envisaged to donate electrons to periplasmic cytochrome  $cd_1$  nitrite reductase (NirS) and membrane-bound nitric oxide reductase (Nor), thus functionally bypassing a cytochrome  $bc_1$ -type complex of proteobacterial denitrifiers (see below) [17].

Several archaeal denitrifiers also possess a NarGH complex (enzyme 1c in Table 2) [18]. In contrast to extant bacterial systems, however, the archaeal NarG protein carries an intact Tat signal sequence suggesting that a cofactor-containing NarGH complex is exported to the potential-positive outside of the cytoplasmic membrane, the p-side [7,19,20]. The genetic context of the corresponding gene clusters implies that NarGH is part of a membrane-bound nitrate reductase system referred to as pNar. The proteins of this putative complex in several haloarchaea such as *Haloferax mediterranei* and *Haloarcula marismortui* 

include a transmembrane diheme cytochrome *b* whose primary structure resembles the cytochrome *b* of cytochrome  $bc_1$  complexes (also termed NarC though unrelated to NarC from T. thermophilus), a Rieske-type iron sulfur protein (NarB) and three other potential membrane proteins including a monoheme cytochrome *b* (Fig. 2C). Therefore, it cannot be excluded that an electrogenic Q-cycle-type mechanism is driven by nitrate reduction coupled to ubiquinol oxidation [19]. The proposed Nar supercomplex might also donate electrons to other denitrifying enzymes such as copper nitrite reductase (NirK), a complex IV HCO-type nitric oxide reductase (cNor) and nitrous oxide reductase (NosZ) (Fig. 2C). An alternative membrane anchor (NarM, a monoheme cytochrome *b* with only two proposed transmembrane domains) was identified in case of the pNar enzyme from the hyperthermophile Pyrobaculum aerophilum [20]. Since the singular heme *b* group is likely to be oriented to the outside of the cytoplasmic membrane, menaquinol oxidation by nitrate is thought to be electroneutral in this organism.

The Nap system of periplasmic nitrate reduction is found almost exclusively in the phylum Proteobacteria. Depending on the species, the Nap system is employed under anoxic and/or (micro)oxic conditions and fulfills several distinct functions, for example in anaerobic respiration or as electron sink during aerobic (photo)organoheterotrophic growth on reduced carbon sources to ensure redox homeostasis, i.e. to dissipate excess reductant [21,22]. Many Proteobacteria encode both Nar and Nap systems whose production is intimately regulated in response to ambient nitrate and oxygen concentrations. Similar to NarG, NapA binds Mo-bis-MGD and a 4Fe-4S cluster. In the majority of known cases NapA forms a complex with the diheme cytochrome c NapB (enzyme 2a in Table 2; see Fig. 3 for models on different Nap systems from representative organisms). Generally, mature NapA is transported across the membrane by the Tat apparatus and this process requires the cytoplasmic chaperone NapD, which is encoded in all known nap gene clusters [23]. At least two independent membrane-bound quinol-oxidizing systems have evolved that are specific to Nap systems (Fig. 3). These are (i) the tetraheme cytochrome c NapC, a member of the NapC/NrfH family (see Section 5.1.4) and (ii) the NapGH complex which consists of two proposed Fe/S proteins (see Section 5.1.2). NapH is a membrane-bound quinol dehydrogenase containing four transmembrane domains while NapG is a periplasmic electron transfer adapter protein. In many, though not all cases, NapGH is accompanied by another iron sulfur protein, NapF, which is thought to bind to NapH at the cytoplasmic side of the membrane (Fig. 3). The function of NapF is unclear but it was hypothesized that it provides reductive power for NapA maturation in the cytoplasm



**Fig. 2.** Model of electrogenic ETCs in Nar-dependent nitrate reduction. A. ETC from formate (formic acid) to nitrate in *E. coli* using the nNar enzyme. B. ETC from NADH to nitrate in *T. thermophilus* using the cNar enzyme. Note that the Nrc and cNar complexes are likely to form a supercomplex. C. Electron transport from quinol to nitrate in haloarchaeal species using the pNar enzyme. See text and Table 2 for details. Electrogenic protons are underlined. Dashed arrows denote speculative reactions. For simplicity, only monomeric enzyme forms are shown. Mo, molybdenum-*bis*-molybdopterin guanine dinucleotide cofactor; Fe/S, iron-sulfur center; *b*, heme *b*.

[24–27]. For *Escherichia coli*, it was shown that NapC acts as a menaquinol dehydrogenase whereas the NapGH complex functions as ubiquinol dehydrogenase [28]. Notably, a *napC* gene is usually absent

from epsilonproteobacterial nap gene clusters [29,30]. Shewanella oneidensis is likely to use CymA, an analogue of NapC, as electron donor to NapAB (see also Section 5.1.4). However, CymA is not specific for the Nap system as it feeds electrons into numerous other respiratory systems including the Nrf system of nitrite reduction (see Section 3.2) [31]. Independently of the architecture of a particular Nap system, it appears that all these assemblies work in a non-protonmotive fashion. In other words, there is no experimental support that Nap-dependent quinol oxidation by nitrate generates a pmf, in contrast to what has been shown or predicted for any of the three Nar systems described above. If true, the main function of the Nap system is nitratedependent regeneration of quinones that are reduced in the course of one or more electrogenic reactions (see Section 4). In addition, the Nap system has been implicated in N assimilation [32,33] and it is not clear whether the quinol oxidation function or the assimilatory function was the primary evolutionary pressure (see Sections 5 and 6).

#### 3.2. Nitrite reductases

Respiratory nitrite reductases are categorized here according to their reaction product, which either is nitric oxide (enzymes 3 and 4 in Table 2) or ammonium (enzymes 5–8) (Fig. 1). Both classes usually reside outside the cytoplasm (e.g. in the periplasm in Gram-negative bacteria or between the cell membrane and the cell wall/S-layer in Archaea). NO-genic nitrite reductases are either cytochromes  $cd_1$ (NirS, containing the unusual cytochrome  $d_1$  at the active site) or copper-dependent enzymes (NirK) that typically contain two copper centers (Cu<sub>I</sub> and Cu<sub>II</sub>). In most cases, these enzymes draw electrons from a periplasmic pool of electron-transferring proteins that consists of (i) various forms of small copper-containing proteins named azurin, pseudoazurin or cupredoxin and (ii) numerous monoheme and diheme cytochromes c like cytochromes  $c_{550}$ ,  $c_{551}$ ,  $c_{552}$  or  $c_{553}$ (Fig. 4) [34]. A complex of Achromobacter xylosoxidans NirK and cytochrome  $c_{551}$  demonstrated the presence of a hydrophobic electron transport pathway at the docking interface of both proteins [35]. The redox mediator pool is reduced by the cytochrome  $bc_1$  complex, which makes electron transport from quinol to nitrite electrogenic. An exception is Pseudomonas stutzeri that is thought to employ the tetraheme cytochrome c NirT protein, a member of the NapC/NrfH family, as the quinol dehydrogenase proposed to directly reduce NirS [36]. In this set-up, it is thought that guinol oxidation by nitrite is electroneutral. Variations of the NirK enzyme were reported that either contained a C-terminal monoheme cytochrome c domain of about 160 to 190 residues or an N-terminal cupredoxin domain (enzymes 4b and 4c in Table 2) [37]. Both domains are likely to serve in electron transfer to the core portion of NirK (Fig. 4B).

The best-known ammonium-generating nitrite reductase is the pentaheme cytochrome c nitrite reductase, NrfA, of which several forms exist (enzymes 5a-d in Table 2) [38]. Depending on the organism, NrfA was described to be a soluble periplasmic protein (for example in the Gammaproteobacteria E. coli and S. oneidensis) or as a subunit of a membrane-bound menaquinol-reactive complex formed by NrfA and a tetraheme cytochrome c of the NapC-type called NrfH (Fig. 5) [39–41]. Such NrfHA complexes are present in Delta- and Epsilonproteobacteria. NrfA catalyzes the six-electron reduction of nitrite to ammonium but commonly also converts alternative substrates like nitric oxide, hydroxylamine, hydrogen peroxide and sulfite [38,42]. This reactive promiscuity of NrfA appears to be beneficial under corresponding stress conditions, indicating that NrfA has a detoxifying function in cell physiology [43-46]. The NrfHA complex was shown to catalyze electroneutral menaguinol oxidation by nitrite in the Epsilonproteobacterium Wolinella succinogenes (Fig. 5A) [39]. In the absence of a NrfH-type quinol dehydrogenase, however, nitrite ammonification in S. oneidensis depends on CymA, the structural and functional analogue of NrfH mentioned above [31]. On the other hand, enteric bacteria like E. coli employ an assembly consisting of the proteins NrfB, NrfC and NrfD to communicate with



Fig. 3. Model of respiratory Nap systems. A. Nap system of *P. pantotrophus*. B. Nap system of *E. coli*. C. The Nap system of *W. succinogenes* which is representative for NapC-lacking Epsilonproteobacteria. D. Nap system of *Desulfovibrio desulfuricans* lacking NapB [201]. See text and Table 2 for details. For further explanations see legend of Fig. 2.

the menaquinol pool (Fig. 5B). Deduced from genetic studies, a membrane-bound menaquinol-oxidizing NrfCD complex was postulated, which is similar to the structurally resolved PsrBC subcomplex of a potential polysulfide reductase from *T. thermophilus* (see Section 5.1.3) [47]. This structure established that a member of the NrfD/PsrC family contains a quinone/quinol binding site near the periplasmic membrane surface [48]. To date, there is no experimental evidence that a member of the NrfD/PsrC family is involved in *pmf* generation. The electron transfer between NrfCD and NrfA in *E. coli* is mediated by the pentaheme cytochrome *c* NrfB, which is distantly related to NrfH (Fig. 5B) [49–51].

Apart from NrfA, several octaheme cytochromes (enzymes 6–8 in Table 2) have been described that, amongst other substrates, convert nitrite to ammonium ([38] and references therein). These cytochromes *c* are part of a taxonomically widely distributed multiheme cytochrome *c* (MCC) superfamily that includes NrfA, hydroxylamine oxidoreductase (Hao) and hydrazine oxidoreductase (Hzo) (see Sections 3.6 and 5.2.1) [52,53]. Especially intriguing is the so-called  $\varepsilon$ Hao (enzyme 8 in Table 2) that is encoded in some epsilonproteobacterial genomes that were described as ammonifiers despite lacking the *nrfA* gene [8,32,54]. Some, but not all,  $\varepsilon$ Hao enzymes are encoded within gene clusters that also encode a member of the NapC/NrfH family.

#### 3.3. Nitric oxide reductases and nitric oxide dismutase

Genomic analyses of microbes implicated in NC transformations have revealed that the molecular diversity of inventory within the nitric oxide conversion module is by far the greatest among all nitrogen cycle modules and this central role of nitric oxide in the nitrogen cycle will be further discussed in Sections 5.2 and 6 [1,55,56]. Respiratory nitric oxide-reducing enzymes either convert nitric oxide to nitrous oxide (Nor; enzymes 10a–d in Table 2) or comproportionate nitric oxide and ammonium into hydrazine (Hzs; enzyme 11) or dismutate two molecules of NO into N<sub>2</sub> and O<sub>2</sub> (Nod; enzyme 12). Moreover, several evolutionarily and structurally unrelated soluble microbial enzymes were reported to convert nitric oxide to either nitrate [flavohemoglobin-NO-dioxygenase (Hmp)], nitrite [hydroxylamine oxidoreductase; cytochrome P460 (CytL)] or nitrous oxide [flavodiiron proteins (Fdp), flavorubredoxin (NorVW), cytochrome  $c_{554}$  (CycA), cytochrome c'-beta (CytS) and cytochrome c'-alpha (CytP)] [45,57–61]. These enzymes are thought to mediate nitrosative stress defense, thus serving primarily in detoxification [44,62]. Notably, cytochrome c nitrite reductase NrfA was reported to reduce nitric oxide to ammonium or nitrous oxide (see Section 3.2), a capability that might be widespread in the MCC family [38,63].

Membrane-bound nitrous oxide-generating NO reductases (Nor enzymes) catalyze nitric oxide reduction at the outside of the cytoplasmic membrane and several such enzymes from denitrifying Proteobacteria, Firmicutes and Archaea have been characterized [64–66]. The best known NO reductases (*c*Nor and qNor; enzymes 10a and 10b in Table 2) either use cytochrome *c*/cupredoxins (Fig. 4A) or quinones as immediate redox partners and both belong to the superfamily of HCOs (see Section 5.2.3). The catalytic site of nitric oxide reduction harbors a dinuclear heme  $b_3$ ::Fe<sub>B</sub> active site that is reduced by another heme *b* group bound by the same protein (named NorB). In *c*Nor enzymes, NorB receives electrons from the monoheme cytochrome *c* subunit NorC while qNor enzymes are quinol-reactive single-subunit enzymes that resemble NorB. High-resolution crystal structures of *Pseudomonas aeruginosa c*Nor and *Geobacillus stearothermophilus* 



**Fig. 4.** Model of ETCs in denitrifying bacteria. A. ETC from ubiquinol to nitrite, nitric oxide and nitrous oxide in *P. denitrificans*. B. ETC from ubiquinol to different NirK-type nitrite reductases. Ubiquinol is reduced by NADH dehydrogenase, succinate dehydrogenase and/or acyl CoA dehydrogenase.  $Cu_Z$  and  $Cu_A$  denote the binuclear  $Cu_A$  and tetranuclear  $Cu_Z$ centres of NosZ while the distinct copper centers of NirK are abbreviated  $Cu_1$  and  $Cu_{11}$ . Heme  $b_3$ ::Fe<sub>B</sub> refers to the dinuclear NO-reducing reaction center of NorBC. Paz, pseudoazurin. See text and Table 2 for details. For further explanations see legend of Fig. 2.

qNor have been reported recently [67,68]. NorB proteins contain 12 transmembrane helices while NorC is anchored to the membrane by a single membrane-spanning segment. In Paracoccus denitrificans, pseudoazurin or cytochrome  $c_{550}$  were found to donate electron to the NorC subunits of a heterotetrameric (NorBC)<sub>2</sub> complex [69]. There is no indication that the cNor complex contributes to pmf generation, i.e. that it takes up protons from the cytoplasmic side of the membrane. The so-called D- and K-channels that serve this function in oxygen-reducing HCOs are absent in cNor enyzmes. Genes encoding "ccNorC" proteins with a second heme-binding domain fused to the usually monohome cytochrome  $c_2$ -type NorC protein have been identified in genomes of Acidovorax sp. JS42, Anaeromyxobacter sp. Fw109-5, Bdellovibrio bacteriovorus HD100, Nitrosospira multiformis ATCC 25196 and Thiobacillus denitrificans ATCC 25259, but no biochemical studies with these proteins have been reported.

In contrast to *c*Nor, qNor enzymes are reactive with ubiquinol and/or menaquinol and contain an N-terminal extension that is absent from NorB in the *c*Nor complex. While this N-terminal extension shows similarity to NorC, a heme *c*-binding motif is lacking. The crystal structure of *G. stearothermophilus* qNor revealed a water channel from the cytoplasm that might serve in proton delivery [68]. Thus, the possibility that qNor might catalyze electrogenic quinol oxidation coupled to nitric oxide

reduction cannot be excluded. An unusual qNor subgroup (Cu<sub>A</sub>-qNor), exemplified by the enzyme from *Bacillus azotoformans* (enzyme 10c in Table 2), contains NorB in a complex with a subunit harboring a Cu<sub>A</sub> site (typically found in oxygen-reducing HCOs), which makes this enzyme competent in receiving electrons from membrane-bound cytochrome  $c_{551}$  in addition to the menaquinol pool [70]. Versions of complex IV HCO-type enzymes with NO reduction activity that differ in critical residues in their active sites have been described (enzymes 10d in Table 2), two of which have been termed sNor (predominantly in nitrogen- and sulfur-catabolic chemolithotrophic bacteria [71]) and gNor (predominantly in sulfur-catabolic chemolithotrophic bacteria ([30,72], see also Section 5.2.3).

Hydrazine synthase (Hzs; enzyme 11 in Table 2) is a key player is the anammox process. Hzs reductively combines nitric oxide (generated by a nitrite reductase) and ammonium to produce hydrazine, which is a central intermediate in the catabolism of anammox bacteria (Fig. 6) [73]. A heterotrimeric complex displaying Hzs activity was purified from the soluble extract of *Candidatus* Kuenenia stuttgartiensis cells [73]. Two of the three proteins (HzsABC, named kuste2859, kuste2860 and kuste2861) are predicted diheme cytochromes *c* and it is thought that the HzsABC complex is located in the anammoxosome lumen, alongside with NirS and hydrazine oxidoreductase (Hzo; see Section 3.6) (Fig. 6). In this scenario, *pmf* generation by anammox bacteria solely relies on the cytochrome  $bc_1$  complex and the electron flow is solely cyclic (see also Section 3.6) [56,73]. The genome of *Candidatus* Scalindua profunda contains a fusion gene that corresponds to the genes encoding kuste2859 and kuste2860 in *Candidatus* K. stuttgartiensis [74].

Finally, a novel nitric oxide-converting enzyme was proposed to act as a crucial enzyme in anaerobic denitrifying methanotrophs [75-77]. These organisms oxidize methane with nitrite as electron acceptor according to the equation  $3 \text{ CH}_4 + 8 \text{ NO}_2^- + 8 \text{ H}^+ \rightarrow 3 \text{ CO}_2 + 4 \text{ N}_2 + 10$ H<sub>2</sub>O. In corresponding enrichment cultures a dominant bacterium belonging to the NC10 phylum was identified and named Candidatus Methylomirabilis oxyfera [78]. This organism is assumed to produce N<sub>2</sub> from nitric oxide (produced from nitrite by either a NirS-type or an εHao-type nitrite reductase) using the suggested enzyme nitric oxide dismutase (Nod, enzyme 12 in Table 2; Fig. 7), thus bypassing nitrous oxide as a denitrification intermediate [76]. The Nod enzyme would produce N<sub>2</sub> and O<sub>2</sub> in equal amounts from nitric oxide (2 NO  $\rightarrow$  N<sub>2</sub> + O<sub>2</sub>). Three quarters of the internally produced oxygen is thought to serve in methane oxidation to CO<sub>2</sub> while the remaining oxygen is probably reduced to water by a cytochrome bo-type quinol oxidase [77]. To date, the nature of the Nod enzyme has not been revealed but it is striking that the genome of *Candidatus* M. oxyfera encodes three putative



Fig. 5. Model of respiratory Nrf systems. A. Nrf system of *W. succinogenes*. B. Nrf system of *E. coli*. See text and Table 2 for details. For further explanations see legend of Fig. 2.



**Fig. 6.** Model of electron transport in the anammox process of *Candidatus* K. stuttgartiensis. The composition of the cytochrome  $bc_1$  complex is speculative. See text and Table 2 for details. For further explanations see legend of Fig. 2. AM, anammoxosome membrane; rET, reverse electron transport.

membrane-bound qNor-like enzymes that are expected to be associated with nitric oxide turnover in this organism [76]. It is assumed that NirS and Nod are not directly involved in *pmf* generation (Fig. 7). Instead, the genome of *Candidatus* M. oxyfera encodes a membrane-bound NADH dehydrogenase (complex I), a cytochrome  $bc_1$  complex (complex III) and several complex IV-type terminal oxidases of the HCO family, thereby employing classic electrogenic electron transport modules [76,77].

#### 3.4. Nitrous oxide reductases

The canonical nitrous oxide reductase of denitrifiers is a homodimeric copper protein designated NosZ (enzyme 13a in Table 2). The periplasmic NosZ receives electrons from the cytochrome *c*/cupredoxin pool in an electroneutral process, thus resembling nitrite and nitric oxide reduction (Fig. 4A). From a bioenergetic point of view, it is therefore not surprising that many denitrifiers reduce nitrate or nitrite only to the level of nitrous oxide due to the absence of a *nosZ* gene [79]. NosZ contains two redoxactive copper centers, termed Cu<sub>A</sub> (dinuclear copper site) and Cu<sub>Z</sub> (tetranuclear copper site) [80–82]. In the functional homodimer, the



**Fig. 7.** Model of respiratory nitrogen metabolism during oxygenic denitrification of nitrite by the methanotroph *Candidatus* M. oxyfera. Oxygen produced by Nod supposedly serves in methane oxidation (75%) as well as in respiratory reduction/detoxification by a cytochrome *bo*-type quinol oxidase (25%). See text and Table 2 for details. For further explanations see legend of Fig. 2.

 $Cu_A$  electron input site of one monomer is in close contact with the catalytic  $Cu_Z$  center of the partner subunit.

The nitrous oxide reductase from Wolinella succinogenes and other Epsilonproteobacteria contains a C-terminally fused monoheme cytochrome *c* domain, which is thought to donate electrons to the Cu<sub>A</sub> site (enzyme 13b in Table 2) [83,84]. Such cNosZ enzymes are encoded in gene clusters that also contain nosG, -C1, -C2 and -H genes which were postulated to encode a putative menaquinol dehydrogenase pathway to *c*NosZ alternative to the conventional cytochrome *bc*<sub>1</sub> complex (Fig. 8A). This pathway comprises a NapGH-type menaquinol dehydrogenase (NosGH, see Section 5.1.2) and two monoheme cytochromes c(NosC1 and NosC2) [84]. Several nos gene clusters from  $\alpha$ -,  $\beta$ - and  $\gamma$ -Proteobacteria encode two FMN-binding flavoproteins (NosR and NosX) that might constitute yet another electron transport pathway from the quinone pool to NosZ (Fig. 8B). NosR resembles NosH but contains an additional periplasmic FMN-binding domain [85]. Despite these variations, no ETC that connects the guinone/guinol pool and NosZ/cNosZ has been reported to be electrogenic.

#### 3.5. Ammonia monooxygenase

The ammonia monooxygenase of aerobic obligate chemolithotrophic bacteria and thaumarchaeota is a complex of three membrane proteins: the heterotrimeric (AmoABC)<sub>3</sub> complex (referred to as Amo in Fig. 9) [86–88]. The active site of the bacterial Amo complex contains nonheme iron and at least one copper center that facilitate the oxidation of ammonia to hydroxylamine, a highly reactive NC that is released into the periplasm and oxidized by the multiheme cytochrome *c* hydroxylamine oxidoreductase Hao (see Section 3.6) [89]. When compared to the bacterial AmoB protein, the thaumarchaeal Amo has a slightly different structure as its AmoB protein, which harbors the catalytic site, lacks one cupredoxin domain and the C-terminal transmembrane spanning domain [90]. In addition, thaumarchaeal genomes do not encode known



**Fig. 8.** Model of alternative electron transport routes to nitrous oxide reductase. A. NosGH-dependent ETC from menaquinol to *c*NosZ in *W. succinogenes*. B. NosR-dependent ETC from quinol to NosZ in *P. stutzeri*. See text and Table 2 for details. For further explanations see legends of Figs 2 and 4.



Fig. 9. Model of ETCs in ammonium-oxidizing bacteria (genus Nitrosomonas; A) and ammonium-oxidizing archaea (genus Nitrosopumilus; B). The red "X" in subpanel B denotes a speculative intermediate which might be hydroxylamine or nitroxyl (HNO). See text and Table 2 for details. For further explanations see legend of Fig. 2.

hydroxylamine detoxifying enzymes and they apparently lack the ability to produce cytochromes *c*. It is thus not clear whether thaumarchaeal Amo produces hydroxylamine or another reactive NC such as nitroxyl (HNO) (Fig. 9B) [90,91]. It has been proposed that the Amo protein complex initiates ammonia oxidation to hydroxylamine when oxygen is activated with two electrons and it has been proposed that these electrons are obtained from the quinol pool whereby electrogenic electron flow was not excluded [89]. However, these authors also stated that there was no experimental evidence for a quinol dehydrogenase function of the Amo complex nor a contribution to *pmf* and to this day, experimental evidence for either proposal is still elusive. Newer models suggest an alternative in form of monooxygenase activation by NO, which is based on observations that relate expression and activity of NO-genic NirK to ammonia monooxygenation in Bacteria [4,92] and Archaea [93,94].

The clustered *amoCAB* genes encoding the subunits of the bacterial Amo complex were believed necessary and sufficient for Amo synthesis and function [86,95–97]; however, the *amoC*, -A and -B genes were recently found as members of co-regulated overlapping operons (amoCABD, amoCAB, amoAB, amoC, amoED), which differ in number [55] and regulation between beta- [98,99] and gammaproteobacterial [100] ammonia-oxidizing bacteria (AOB). In Beta-AOB, the amoD gene (which encodes a potentially monotopic membrane protein of unknown function) is found in tandem with a likely duplicated orthologue (amoE) downstream of the amoCAB genes [101] and first expression experiments suggest that amoED is co-regulated and not part of the same operon as amoCAB [102]. In Gamma-AOB, the amoD gene is expressed only as a member of the amoCABD operon and significantly higher transcript levels of amoC as compared to amoCABD and amoAB are achieved by differential regulation (initiation and termination) of the amoCABD gene cluster [100]. In contrast, genomes of Beta-AOB encode multiple copies of non-operonal amoC genes [55,97] that have been implicated in recovery from ammonia starvation [102]. All AOB encode also amoD

singletons [101] whereas singleton amoA and amoB genes have not yet been found in any AOB genome. Interestingly, orthologues of amoD (but not *amoE*) are also encoded by genomes of aerobic nitrifying methanotrophs where they reside either downstream of the gene cluster encoding particulate methane monooxygenase (pMmo) in Alphaproteobacteria (type-II methanotrophs) or in proximity of a gene tandem encoding blue copper oxidases in Gammaproteobacteria (type-I and -X methanotrophs). Homologues of bacterial amoD and amoE genes have only been found in genomes of nitrifying bacteria. While it is known for quite some time that the Amo and pMmo complexes are encoded by homologous genes [103], it is only clear since recently that both enzymes are members of a superfamily of copper-dependent membrane monooxygenases (Cu-MMOs) with significant promiscuity for reduced carbon and nitrogen compounds (see Section 5.2.4). Based on this homology, it is likely that ammonia oxidation facilitated by Amo operates identical to methane oxidation facilitated by pMmo, whose crystal structure has been resolved and whose biochemistry has thus been extensively studied [104,105]. To this day, Amo has not been crystallized.

The thaumarchaeal Amo differs from its bacterial homologue not only in structure and, maybe, in function, its genetic basis is also significantly different. While detailed *amo* gene expression studies are not yet available, the organization of thaumarchaeal *amoA*, *amoB* and *amoC* genes in the genome varies between individual thaumarchaeal lineages and they are not clustered in the canonical bacterial order *amoCAB* or are present in more than one copy [90,91,106].

#### 3.6. Hydroxylamine and hydrazine oxidoreductases

The capacity of AOB to aerobically catabolize ammonia as their sole source of energy and reductant requires another specialized protein complex in addition to Amo: hydroxylamine oxidoreductase (Hao) as well as redox-active cytochromes c, which relay the electrons extracted from hydroxylamine to the quinone pool (Fig. 9A) [86,89,107]. The oxidation of hydroxylamine to nitrite is catalyzed in the periplasm by Hao, which consists of three octaheme cytochrome c monomers (HaoA) cross-linked between a tyrosine residue and the active site heme c group of the neighboring monomer [89,108]. The circular symmetric arrangement of the HaoA<sub>3</sub> complex is regarded as a prerequisite for the oxidative chemistry of the enzyme [109,110]. Hao lacking the critical tyrosine residue failed to oxidize hydroxylamine to nitrite [89].

Based on experiments that evidenced in vitro reduction of cytochrome  $c_{554}$  (CycA) by Hao [111], modeling of this interaction [112,113] as well as demonstration of guinone reductase function of cytochrome  $c_{m552}$  (CycB) [50], an early linear model of electroneutral electron flow from hydroxylamine to quinone (Hao  $\rightarrow c_{554} \rightarrow c_{m552}$ ) had been established (Fig. 9A). The observed clustering of Hao- and cytochrome  $c_{554}$ - and  $c_{m552}$ -encoding genes in genomes of all AOB [55] along with the proposed interaction of their products led to a designation of the "Hydroxylamine Ubiquinone Redox Module" (HURM), even though a redox interaction between cytochromes  $c_{554}$  and  $c_{m552}$ in vitro or during ammonia-oxidation to nitrite in AOB as well as the functionality of the redox chain in the absence of either cytochrome has never been experimentally established [56]. Therefore, a direct interaction of HaoA and cytochrome  $c_{m552}$  cannot be excluded (Fig. 9A). Cytochrome  $c_{m552}$  is related to the NapC/NrfH QRP family (see Section 5.1.4) and it is therefore likely that quinone reduction is achieved using protons taken up from the periplasmic space. If true, this would make HURM-catalyzed hydroxylamine oxidation by quinone an electroneutral process (Fig. 9A). Interestingly, cytochrome c<sub>554</sub> has a mild NO reductase activity although it is unclear whether this is physiologically relevant [61]. Nevertheless, clustering of cytochrome  $c_{554}$  (*cvcA*)-like genes with homologs of the *napC/nrfH* gene family in genomes of Bacteria that are not AOB suggest potential for such interaction in form of a quinol-reactive nitric oxide reduction module [114].

It is significant to mention that HURM, which contains a ORP as an essential module element, has been identified thus far only in those nitrifying bacteria that can support growth based on catabolism of ammonia as the sole source of energy and reductant. All other nitrifying bacteria such as methane-oxidizing Proteobacteria and Verrucomicrobia lack HURM (in particular, a homologue of *cycB*) and they need additional sources of energy and reductant [114]. The ammonia-based HURM genes are organized in a conserved gene cluster, haoAB-cycAB [51,114], however as in the case of Amo-encoding genes, HURM-encoding genes are differentially expressed in Beta- and Gamma-AOB. Previous studies with the Beta-AOB *N. europaea* suggested that the *haoA* gene and the *cycAB* genes are expressed independently, and no evidence for the transcription of haoB (orf2) was found [115,116]. In contrast, studies of the transcriptional response of the Gamma-AOB such as Nitrosococcus oceani ATCC 19707 to ammonia suggested the presence of a steady-state mRNA that included all four genes; nevertheless, basal expression produced independent haoAB and cycAB transcripts [117]. Interestingly, ammonia and hydroxylamine, although both sources of energy and reductant, differentially induced gene expression in AOB including amo and hao genes [118]. Ammonia also induced expression of an haoAB gene tandem in the nitrifying methanotroph, Methylococcus capsulatus Bath [119], which provided the initial designation of the first two genes in the HURM gene cluster as haoAB despite thefact that M. capsulatus lacks cycAB genes and thus HURM. Analysis of haoA gene expression in N. europaea strain ENI-11 revealed differential regulation and identified the one haoA gene copy not located in the vicinity of the two amoCAB operons as being expressed at the highest level and as the sole haoA gene in cells denied an energy source [120].

Anaerobic ammonia-oxidizing (anammox) bacteria also harbor significant Hao activity and proposals in the pre-genomics era suggested that Hao participated in the anammox process with a dual function by also oxidizing hydrazine [121,122]. The first genome sequence of an anammox bacterium did not only lead to a correction of the initially proposed biochemistry of the anammox process, which established NO as the critical intermediate, but also unraveled a wealth of unexpected inventory with dedicated functions in hydrazine synthesis and oxidation as well as nitrite and nitric oxide transformations (Fig. 6) [123–125]. Ongoing genome analysis and subsequent biochemical and physiological experiments confirmed the production of active Hao enzymes and it turned out that only one of the many encoded octaheme cytochromes c encoded in the Candidatus K. stuttgartiensis genome (kustec0694; named hydrazine oxidoreductase or Hzo; enzyme 16 in Table 2) was dedicated to hydrazine oxidation; an enzyme sensitive to hydroxylamine and nitric oxide [73]. However, it is unknown how electrons derived from hydrazine oxidation ultimately reach the cytochrome  $bc_1$  complex and whether this process is electrogenic, e.g. by involving a quinone species (Fig. 6). This would leave the cytochrome *bc*<sub>1</sub> complex as the main coupling site in anammox metabolism. However, the Candidatus K. stuttgartiensis genome encodes three different cytochrome *bc*<sub>1</sub>-type complexes whose biochemical properties await investigation (e.g. with respect to quinone reactivity) [123]. One of those (kuste4569-4574) was reported to be highly expressed and was speculated to function as a pmf-generating complex in the anammoxosome membrane [73]. It is noteworthy that the monoheme cytochrome  $c_1$  of Proteobacteria might be functionally replaced by di-, hexa- or octaheme cytochromes c in different Candidatus K. stuttgartiensis bc1-type complexes.

In contrast to *hao* genes in aerobic AOB, most genes encoding functional Hao and Hzo proteins in anammox bacteria are not clustered with genes encoding other cytochromes *c* or proteins that have been implicated in other redox and electron flow functions [73]. However, specific genetic differences provided tools for discriminating detection of *hzo* and *hao* genes in the environment, a difficult approach because *hzo* and *hao* genes are homologous [52,126]. Both Hao and Hzo belong to the MCC family already mentioned in Section 3.2. The evolutionary history of the MCC family will be discussed in Section 5.2.1.

#### 3.7. Nitrite oxidoreductase

Early work with nitrifying bacteria unraveled the critical function of a membrane protein complex termed nitrite oxidoreductase in obligate aerobic Proteobacteria able to grow on energy derived from the oxidation of nitrite to nitrate [127]. Initial genetic and biochemical experiments identified that these specialized nitrifiers required fairly high substrate concentrations due to the need for nitrite uptake and that the nitrite oxidoreductase complex was part of a simple respiratory electron flow involving an electrogenic cytochrome *c* oxidase HCO of the aa3-type (Fig. 10A) [128]. Later discoveries of other aerobic nitrite oxidizers with different nitrite oxidation kinetics and taxonomic affiliations led to the discovery of new inventory, a clean-up in gene terminology (establishing Nxr as the designated term for nitrite oxidoreductase) and mind-boggling evolutionary relationships between this inventory (assumed to specifically operate in obligate aerobic bacteria) and the NC-transforming inventory in anammox bacteria [11,129–134]. It has been proposed for Nitrobacter species that nitrite is oxidized by a membrane-bound Nxr complex consisting of four subunits (NxrABIC; Fig. 10A; enzyme 9a in Table 2) in an arrangement similar to the cNar complex (Fig. 2B) [132,133]. The NxrAB assembly contains molybdenum and iron and is thought to be located at the cytoplasmic surface of the membrane (therefore called nNxr here). NxrI is a membrane-bound diheme cytochrome *b* and NxrC is a diheme cytochrome *c*, which might act as an electron donor to the terminal oxidase (Fig. 10A). Given the positive standard potential of the nitrate/nitrite redox pair, it is unlikely that a quinone is involved in the respiratory ETC from nitrite to oxygen or that this process is electrogenic. Cytoplasmic nitrite oxidation necessitates nitrite import and nitrate export, both of which might be catalyzed by specific transporters of the proteobacterial NarK-type. Furthermore, a reversed ETC from the guinone pool (possibly reduced by nitrite through NxrI) to NADH via cytochrome  $c_{550}$ , the cytochrome *bc*<sub>1</sub> complex and a NADH:ubiquinone oxidoreductase (complex I) is predicted by the inventory of Nitrobacter genomes [132,133]. A guinone reductase activity of NxrI could also explain why some Nitrobacter species are able to grow anaerobically with nitrate as electron acceptor [135]. Produced nitrite is thought to ultimately end up in nitrous oxide trough the catalysis by NirK and a so far unidentified nitric oxide reductase (or by sNOR as found in *Nitrobacter hamburgensis* [133]).

In *Candidatus* Nitrospira defluviii (phylum Nitrospirae), the nitrite-oxidizing site of the NxrAB assembly is likely to be oriented



Fig. 10. Model of ETCs in different nitrite-oxidizing bacteria. A. Bacteria of the genus *Nitrobacter* (phylum Proteobacteria). B. Bacteria of the genus *Nitrospira* (phylum Nitrospirae). See text and Table 2 for details. For further explanations see legend of Fig. 2. rET, reverse electron transport.

to the outside of the cytoplasmic membrane (pNxr; enzyme 9b in Table 2; Fig. 10B) [11]. In this case, it is unknown how electrons are transported to the oxygen-reducing enzyme (probably a terminal oxidase of the cytochrome bd-type) or fed into the reverse ETC (Fig. 10B). The involvement of cytochrome(s) b and c cannot be excluded [11].

Fueled by early suggestions that proteobacterial nitrite oxidoreductase (NxrAB) and nitrate reductases (NarGH) have close genetic relationships and similar inhibitor profiles [136], recent in silico and experimental studies revealed that the Nxr complex in anammox bacteria (aNxr; enzyme 9c in Table 2; Fig. 6) facilitates anaerobic nitrite oxidation to nitrate, which provides energy and reductant needed for growth [73,74]. Similar to the closely related Nxr enzyme in Candidatus N. defluviii [11], the extracted electrons could flow via cytochromes to the quinol pool [123,124]. It has been proposed that the Nxr complex in anammox bacteria may also function reversely as a true nitrate reductase when oxidizing small organic molecules (e.g. proprionate) with nitrate as electron acceptor [137]. The genes proposed to encode the required soluble complement of monoand diheme cytochromes c have been identified in Candidatus K. stuttgartiensis and in Candidatus S. profunda [74,134]. The phylogenetic relationship of the NarGH-NxrAB family is discussed in Section 5.2.2.

### 4. A comparative view on the design of bioenergetic systems in respiratory nitrogen metabolism

Despite the numerous differences concerning individual enzymes and electron transport pathways displayed in Figs 2-10, Section 3 has revealed some common elements of the bioenergetic systems in question. The most important fact is that almost all of them appear to depend on the presence of (i) quinones/quinols and (ii) cytochromes b and/or c. Furthermore, iron-sulfur centers are obligatory to the vast majority of the discussed metabolic processes and many different fusion proteins have been found that originated from the genetic combination of otherwise separate electron transfer modules like monoheme cytochromes *c* and cupredoxins. With the exception of the nNar, cNar and nNxr systems, NC conversion takes place at the outside of the cytoplasmic membrane such as in the periplasm or in the lumen of stacks of intra-cytoplasmic membranes of Gram-negative bacteria (Proteobacteria, members of the NC10 phylum), in the anammoxosome of the Brocadiaceae or between the plasma membrane and the cell wall/S-layer in Archaea. Proven mechanisms of *pmf* generation discussed in Section 3 are the redox loop of nNar/cNar nitrate reductase complexes and the Q-cycle that operates in cytochrome  $bc_1$  complexes and alternative complex III-type structures (see Section 5.1.3) [138]. Additional protonmotive steps are frequently present, especially in microorganisms that grow chemoorganotrophically on carbohydrates, lipids, fatty acids, proteins and/or amino acids or by aerobic respiration (with externally present or internally produced dioxygen). Such cells usually contain proton-pumping enzyme complexes, such as NADH:quinone oxidoreductase and electrogenic HCOs, or redox loop enzymes other than Nar-type nitrate reductases [6]. A prominent example is the membrane-bound formate dehydrogenase complex of enteric bacteria that oxidizes formate, a common bacterial fermentation product, to yield carbon dioxide, a proton and two electrons (Fig. 2A). The architecture of this enzyme resembles that of nNar but with an "inverse" orientation resulting in periplasmic formate oxidation [13]. Therefore, protons are released in the periplasm and this process is coupled to proton uptake in the course of menaquinone reduction taking place at the cytoplasmic surface of the diheme cytochrome *b* membrane anchor subunit FdhC [6]. As with nNar, the  $H^+/e^-$  ratio of this mechanism is one. Heterotrimeric membrane-bound Ni/Fe-hydrogenases work according to the same principle [6,139–141].

Taken together, Nar complexes, i.e. enzymes 1a, 1b and 1c in Table 2, appear to be the only experimentally proven NC-transforming enzymes that couple substrate conversion directly to *pmf* generation (Table 3; see footnotes of Table 3 for possible exceptions to this rule). As described above and considering evolutionary history, the nNar and cNar enzymes are thought to carry out a relatively recently invented redox loop mechanism due to the conversion of NarGH orientation from the outside to the inside of microbial cells (possibly combined with membrane attachment and the resulting direct link to quinol oxidation). All other bioenergetic systems presented in Table 3 rely on protonpumping, the Q-cycle or redox loop enzymes that operate independent of, but are driven by, NC conversion. In these cases, maybe surprisingly, NCs serve as non-protonmotive electron sources or sinks or, in other words, in electroneutral redox balancing of, for example, the quinone/quinol pool. The Nap system of nitrate reduction and its versatile function in microbial physiology is a prime example for this fact (see Section 3.1). Since electroneutrally operating systems are unable to conserve the free energy of a redox reaction (which rather is lost as heat) and assuming that the non-protonmotive character of particular redox reactions was preserved during evolution, it appears that *pmf* generation (and hence ATP generation) was not a driving force during the evolution of such bioenergetic systems. In

#### Table 3

Bioenergetic coupling of electron transport chains involved in respiratory NC transformation.

Metabolic property and corresponding nitrogen compound transforming enzyme <sup>a</sup>	Involved enzymes or modules ( <i>pmf-generating</i> modules in bold)	Is NC transformation directly coupled to <i>pmf</i> generation? <sup>b</sup>
<ol> <li>Nitrate reduction by formate; nNar (1a)</li> <li>Nitrate reduction by NADH; cNar (1b)</li> <li>Nitrate reduction by quinol; pNar (1c)</li> <li>Nitrate reduction by quinol; NapA (2a, 2b)</li> <li>NO-genic nitrite reduction by quinol; NirS or NirK (3, 4)</li> <li>Ammonifying nitrite reduction by quinol; members of the MCC family (5–8)</li> <li>N<sub>2</sub>O-genic nitric oxide reduction by quinol; cNor (10a)</li> </ol>	FDH complex (FdhGHI) $\rightarrow$ Q/QH <sub>2</sub> $\rightarrow$ nNar (NarGHI) NADH dehydrogenase $\rightarrow$ Q/QH <sub>2</sub> $\rightarrow$ cNar (NarGHIC) QH <sub>2</sub> $\rightarrow$ pNar QH <sub>2</sub> $\rightarrow$ NapC/NapGH $\rightarrow$ (NapM) $\rightarrow$ NapA/NapAB QH <sub>2</sub> $\rightarrow$ cytochrome $bc_1$ <sup>d</sup> $\rightarrow$ ETP $\rightarrow$ NirS/NirK QH <sub>2</sub> $\rightarrow$ Member of the NapC/NrfH family or NrfBCD $\rightarrow$ cytochrome $c$ nitrite reductase of the MCC family QH <sub>2</sub> $\rightarrow$ cytochrome $bc_1 \rightarrow$ ETP $\rightarrow$ cNor	Yes (redox loop; Fig. 2A) Presumably yes (redox loop; Fig. 2B) Presumably yes (potential Q-cycle; Fig. 2C) No <sup>c</sup> (Fig. 3) No (Fig. 4) No <sup>e</sup> (Fig. 5) No (Fig. 4A)
8. N <sub>2</sub> O-genic nitric oxide reduction by quinol; qNor (10b) 9. N <sub>2</sub> -genic nitrous oxide reduction by quinol; Nos/cNos (13a, 13b)	$QH_2 \rightarrow qNor$ $QH_2 \rightarrow ETP/NosGH/NosR(X) \rightarrow Nos/cNos$	Presumably <b>no</b> <sup>f</sup> <b>No</b> (Figs. 4A and 8)
10. Anammox reaction; NirS, Hzs, Hzo (3, 11, 16) 11. Ammonium oxidation by oxygen; Amo, Hao (14, 15)	$QH_2 \rightarrow cytochrome \ bc_1 \rightarrow ETP \rightarrow NirS/Hzs/Hzo \rightarrow ETP \rightarrow Q$ AmoABC $\rightarrow Q \leftarrow cytochromes \ c \leftarrow Hao; \ QH_2 \rightarrow cytochrome \ bc_1 \rightarrow Oxidase$	<b>No</b> (Fig. 6) <b>No</b> (Fig. 9A)
12. Nitrite oxidation by oxygen; NxrAB (9a, 9b)	NxrAB $\rightarrow$ cytochrome <i>b</i> (and <i>c</i> ) $\rightarrow$ <b>Oxidase</b>	<b>No</b> (Fig. 10)

ETP, electron transfer proteins forming a redox mediator pool. These proteins comprise monoheme cytochromes c and/or small copper redox proteins (cupredoxins).

<sup>a</sup> Numbers in parentheses refer to enzyme designations given in Table 2.

<sup>b</sup> See Sections 3 and 4 for details.

<sup>c</sup> Unless the NapGH complex catalyzes electrogenic quinol oxidation (see Section 5.1.2).

<sup>d</sup> Exceptionally, NirT replaces the cytochrome  $bc_1$  complex in *P. stutzeri* which would make this step electroneutral.

<sup>e</sup> Unless the NrfCD complex catalyzes electrogenic quinol oxidation (see Section 5.1.3).

<sup>f</sup> Unless qNor is a proton pump (see Section 3.3).

accordance with this assumption, it seems that bioenergetic systems are not optimized to obtain maximal  $H^+/e^-$  ratios.

It was recently suggested that selection for high-throughput redox systems that generate reactive and toxic intermediates was highly unlikely unless the genome already encoded significant detoxification capacity [142]. Following this line of thought, it is likely that many of the extant respiratory enzymes primarily served in detoxification of harmful NCs described in Table 1 (see also Section 6). Later on, higher concentrations of these substances and lateral acquisition of high-throughput production modules might have favored the adoption of such enzymes in catabolic metabolism in order to facilitate the function of already present protonmotive, i.e. ATP-producing mechanisms (like membrane-bound hydrogenases and formate dehydrogenases, NADH dehydrogenases, cytochrome complexes of the *bc*-type and terminal HCO-type oxidases).

### 5. The phylogeny of electron transport modules essential to bioenergetic systems of NC conversion

From the previous sections it is inferred that NC conversion depends on a finely tuned arrangement of suitable electron transport modules that likely have been interchanged during evolution of extant ETCs. In this section, knowledge on some prominent modules is reviewed and the phylogeny of distinct protein families is discussed.

#### 5.1. QRP families

As described above, the recruitment of an NC converting enzyme is insufficient to drive a bioenergetic system unless the enzyme is connected to an electron transport network that commonly involves the quinone/quinol pool. It is generally agreed upon that quinones (in particular low-potential menaquinones), cytochromes (predominantly cytochromes b and c) and Fe/S cluster proteins have evolved early in the history of life [143-145]. Interestingly, functional analogues of evolutionary progenitors of the quinone-reactive cytochrome  $bc_1$  complex are also thought to represent ancient protein assemblies that have always been functionally located at the core of bioenergetic systems [56]. Examples are a highly expressed gene cluster encoding a putative complex III in the anammoxosome membrane (proteins kuste4569-4574) ([73,123] and references therein) or the alternative complex III that evolved from the NrfD/PsrC-type QRP module (see Section 5.1.3) [138] Therefore, one of the key questions regarding the formation of NC-converting systems is how QRPs have evolved in order to create the versatile cyclic and/or linear electron transport-facilitating modules that they represent today. It is indisputable that genes encoding ORP modules have been shuffled during evolution to create new genetic contexts (module tinkering) and that these have probably been transferred extensively between individual genomic entities by horizontal gene transfer. Abstraction of bioenergetic systems involved in microbial NC transformations reveals four important QRP modules: the diheme cytochromes b, the NapH/NosH and NrfD/PsrC protein families and the NapC/NrfH family comprising several subfamilies of multiheme cytochromes c (see Table 4 for a compilation). Interestingly, the era of genomics has afforded us to recognize that these QRPs exist both as stand-alone modules and as parts of bioenergetically relevant protein complexes. The combination of QRPs was likely driven by both the evolution of the quinone pool as well as some prominent enzyme families involved in NC turnover, which will be discussed in Section 5.2.

#### 5.1.1. Quinone/quinol-reactive diheme cytochromes b

Membrane-bound diheme cytochrome *b* QRPs are key components of several enzyme complexes serving in aerobic and anaerobic respiratory ETCs of many bacterial and archaeal species although detailed information on them usually derived from studies with proteobacterial model organisms. The reactivity of such cytochromes with both menaquinone and the evolutionary younger ubiquinone

#### Table 4

Compilation of Qrp families relevant to this article. Note that QRPs of the first three families appear to be solely reactive with iron-sulfur proteins. Families 2 to 4 are independent of cytochrome *b*.

Qrp family and typical members. The corresponding respiratory system is given in parentheses.	Redox partner protein (and its cofactor) other than quinone/quinol
1. Diheme cytochromes <i>b</i> 1.1 Cytochrome <i>b</i> of the cytochrome <i>bc</i> <sub>1</sub> complex	Rieske protein (Fe/S) and cytochrome $c_1$
1.2 FdhC/HydC (formate dehydrogenase/hydrogenase)	FdhAB (MGD, Fe/S)/HydAB (Ni/Fe, Fe/S)
1.3 Narl (nitrate reductase) 1.4 NarC (haloarchaeal nitrate reductase)	NarGH (MGD, Fe/S) NarB (?) (Fe/S) (?)
2. NapH/NosH family	
2.1 NapH (nitrate reductase)	NapG (Fe/S)
2.2 NosH (nitrous oxide reductase)	NosG (Fe/S)
2.3 NosR (nitrous oxide reductase) 3. NrfD/PsrC family	Not known
3.1 NrfD (cytochrome <i>c</i> nitrite reductase)	NrfC (Fe/S)
3.2 PsrC (polysulfide reductase)	PsrB (Fe/S)
3.3 TtrC (tetrathionate reductase)	TtrB (Fe/S)
3.4 MccD (cytochrome <i>c</i> sulfite reductase)	MccC (Fe/S)
3.5 ActC (alternative complex III)	ActB (Fe/S)
3.6 QrcD (quinone reductase complex from sulfate reducers)	QrcC (Fe/S)
<ol> <li>Multiheme cytochromes c of the NapC/NrfH family</li> </ol>	
4.1 NapC (nitrate reductase)	Multiheme cytochromes
	<i>c</i> NapB (2 heme <i>c</i> ) or
	NapM (4 heme $c$ )
4.2 NrfH (cytochrome <i>c</i> nitrite reductase)	NrfA (5 heme $c$ )
4.3 NirT (cytochrome $cd_1$ nitrite	Cytochrome <i>cd</i> <sub>1</sub>
reductase)	(hemes $c$ and $d_1$ )
4.4 CymA (multifunctional)	Various heme c proteins
4.5 Cyt. <i>c</i> <sub>m552</sub> (hydroxylamine oxidoreductase)	Cytochrome $c_{554}$ (4 heme $c$ )
4.6 FccC (methacrylate reductase)	FccB (4 heme $c$ )
4.7 TorC <sup>a</sup> (TMAO reductase)	TMAO reductase TorA
	(MGD, Fe/S)
4.8 DorC <sup>a</sup> (DMSO reductase)	DMSO reductase DorA (MGD, Fe/S)

<sup>a</sup> Fusion proteins between a core tetraheme cytochrome *c* and a C-terminal monoheme cytochrome *c* domain.

has been shown and many guinone-binding sites have been structurally resolved. Prominent examples are the cytochrome  $bc_1$  complex as well as guinone-reactive membrane anchor subunits of nNar, cNar, Ni/Fe-hydrogenase, formate dehydrogenase, succinate dehydrogenase and fumarate reductase (Table 4) ([6] and references therein). In addition, such cytochromes b also occur in cytochrome bo<sub>3</sub> and cytochrome bd quinol oxidases. The two heme b groups are located within the membrane plane, usually oriented to different sides of the membrane and axially ligated by histidine side chains that are provided by at least two different transmembrane helical regions (note that the distribution of histidine ligands is not conserved [139,146]). Redox partner proteins of diheme cytochromes b are mainly iron-sulfur proteins or *c*-type cytochromes (Table 4; see also Figs 2 and 4 for examples). In many cases, electron transport catalyzed by diheme *b*-type cytochromes is coupled to the generation of a *pmf* across the membrane, either by proton pumping, the Q-cycle or the redox loop mechanism (Table 3) [6].

#### 5.1.2. Quinol dehydrogenases of the NapH/NosH family

Prototypic NapH/NosH-type proteins contain four hydrophobic segments that are likely to form transmembrane domains, two conserved CX<sub>3</sub>CP signatures and two four-cysteine clusters typically involved in [4Fe-4S] center formation. The NosR protein mentioned in Section 3.4 also belongs to this family (Table 4). Prototypic NapH/NosH proteins are thought to form stable complexes with specific periplasmic redox partner proteins called NapG or NosG that are predicted to be polyferredoxins with four [4Fe-4S] iron-sulfur centers (Figs 3B-D and 8). Recently, the formation of a NapGH complex has been shown in the case of W. succinogenes [147]. NapGH/NosGH complexes are assumed to catalyze ubi- or menaquinol oxidation and to transfer electrons to cognate periplasmic *c*-type cytochromes in electron transport chains leading to either periplasmic nitrate reductase (NapA) or cytochrome c nitrous oxide reductase (cNosZ) (see Sections 3.1 and 3.4). The expression of chimeric nap/nos operons in W. succinogenes indicated that a NapG-NosH complex is functional in nitrate respiration whereas NosG was apparently not capable of interacting with the NapAB system when expressed in either a nosGH or nosG-napH arrangement [147]. The NapG-NosH assembly was found to be less stable than the NapGH complex since NapG could be easily washed off the membrane by applying low concentrations of sodium chloride. Thus, it seems conceivable that NapG and NosG represent adaptor proteins that determine the specificity of periplasmic electron transport to the corresponding terminal reductase.

The orientation of the four transmembrane segments of NapH was determined experimentally for *E. coli* NapH using a series of alkaline phosphatase and  $\beta$ -galactosidase fusion proteins and implying that all four poly-cysteine motifs are located at the cytoplasmic side of the membrane [148] where they might interact with a cytoplasmic protein like NapF [27] (see Section 3.1). The architecture and localization of the quinol-reactive site in NapH/NosH is not known and it cannot be excluded that more than one such site is present. Furthermore, the function of the cytoplasmic, potentially metal-binding CX<sub>3</sub>CP motifs and the putative iron-sulfur centers has not been elucidated. The potential participation of NapH/NosH-type proteins in *pmf* generation has not been examined.

Interestingly, *napH*-like genes were also discovered in genetic contexts other than nitrate or nitrous oxide reductase gene loci. These genes are *mauN*, *rdxA* and *ccoG* that are present in methylotrophs (methylamine dehydrogenase gene cluster), photosynthetic and aerobic bacteria (cytochrome *cbb*<sub>3</sub> oxidase accessory gene cluster) respectively [48].

#### 5.1.3. Quinol dehydrogenases of the NrfD/PsrC family

NrfD is a polytopic membrane protein encoded in the *nrf* gene cluster of enteric bacteria. It was shown to be essential for formate-dependent nitrite reduction of *E. coli* and was predicted to traverse the membrane eight times with both the N- and C-termini located in the periplasm [149]. NrfD forms a complex with NrfC, which is predicted to bind four [4Fe-4S] clusters (Fig. 5B). The NrfD homologue PsrC was experimentally shown to form a complex with the NrfC-type protein PsrB and the catalytic subunit PsrA in the membrane-bound polysulfide reductase complex of W. succinogenes [150]. PsrC is the membrane anchor of the complex and the two hydrophilic subunits PsrB and PsrA are oriented toward the periplasm. The crystal structure of a NrfD/PsrC-type QRP from T. thermophilus confirmed the predicted presence of eight transmembrane helices and was found to contain a single quinol-binding site at the periplasmic membrane surface [47]. Comparative studies revealed that NrfD-type proteins are widespread in bacterial anaerobic electron transport chains with homologous proteins present, for example, in polysulfide, tetrathionate and sulfite reductase systems (Table 4). Furthermore, such proteins were reported as subunits of "alternative complex III" (ActC) and a quinone reductase complex from sulfate reducers (QrcD) [151,152]. The corresponding genes are found in a variety of different genetic contexts, usually in combination with a gene encoding a polyferredoxin similar to NrfC/PsrB.

#### 5.1.4. The NapC/NrfH family of multiheme cytochromes c

The proteins of this family are membrane-bound tetra- or pentaheme cytochromes c that comprise an N-terminal membrane-spanning helix and a globular cytochrome c domain situated at the outside of the bacterial membrane (Table 4) [153]. Most members are tetraheme cytochromes c consisting of about 175 amino acid residues while some

proteins of the TorC clade contain a C-terminal monoheme cytochrome *c* domain that very likely resulted from a gene fusion [154]. The NapC/NrfH family was previously referred to as the NapC/NirT family since NirT from P. stutzeri was the first such protein to which a function had been assigned (see Section 3.2) [36]. Within their respective functional modules, NapC and NrfH donate electrons, either directly or indirectly, to periplasmic nitrate reductase (NapA) and periplasmic cytochrome *c* nitrite reductase (NrfA), respectively, whereas another family member, cytochrome  $c_{m552}$ , was proposed to feed electrons derived from hydroxylamine oxidation into the ubiquinone pool of nitrifying organisms ([50,51,155–157]) and references therein) (Fig. 9A). In each case, multiheme cytochromes c act as redox partner proteins (Table 4). In Shewanella species, CymA acts as an electron transport hub in several respiratory chains (see Section 3.1). Recently, CymA from Shewanella oneidensis was functionally characterized and found to contain a bound menaquinone molecule [158,159]. The crystal structure of NrfH from Desulfovibrio vulgaris was solved as part of a quinol oxidizing (NrfHA<sub>2</sub>)<sub>2</sub> complex [40,160]. A menaguinol binding site was shown to be present at the periplasmic side of the membrane, situated in the vicinity of heme 1, which serves as an entrance to a densely packed electron-transferring heme *c* wire [156]. It is therefore very likely that protons are exchanged with the periplasmic space during guinone/guinol turnover and this would rule out the participation of NapC/NrfH-type and  $c_{m552}$ -type cytochromes in *pmf* generation and consumption, respectively (Fig. 5A and Table 3). Notably, the axial heme *c* ligands of the NrfH and NapC families differ considerably [157,161].

Prior phylogenetic analyses of proteins in the NapC/NrfH QRP superfamily already reported relationship between the membrane-bound QRP, NrfH, and the soluble periplasmic NrfB protein, the pentaheme cytochrome *c* protein that relays electrons from the NrfCD complex to the catalytic NrfA protein in formate-oxidizing ammonifying proteobacteria (see Section 3.2) [51]. Unexpectedly, a recent analysis revealed that members of the NapC/NrfH QRP superfamily may interact with a great diversity of functional redox partners and that the explosion over evolutionary time into the diversity of the extant superfamily likely started from the NrfH-type ([51] and our unpublished results). While extant NrfH and NapC appear to interact solely with menaguinol in predominantly anaerobic bacteria and cytochrome  $c_{m552}$  interacts solely with ubiquinone in obligate aerobic bacteria (note the reversal of electron flow direction), NirT and TorC are assumed to interact with both guinol types in aerobic, facultative anaerobic and anaerobic bacteria. It is thus most parsimonious to predict that an ancestral tetraheme cytochrome *c* ORP-encoding *nrfH*-type gene has emerged in a sulfur-, formaldehyde- or hydrogen-oxidizing anaerobic respiratory bacterium.

#### 5.1.5. Concluding remark on the presented QRP families

The data presented in Sections 5.1.1 to 5.1.4 (see also Table 4) suggest the evolution of four independent QRP families based on common respiratory cofactors such as heme *b*, heme *c* and iron-sulfur centers. In general, the functional crosstalk between members of different QRP modules is assumed to be scarce despite the fact that an exceptional interaction of NapH and NapC has been reported in the *E. coli* Nap system [148]. All four QRP modules seem to be early inventions during evolution (see also Section 6) although the origin of individual QRP families remains elusive.

#### 5.2. Enzyme families involved in nitrogen compound turnover

Similar to the QRP families discussed above, Section 3 has revealed the existence of enzyme (super)families involved in NC transformation, the most prominent of which are presented in Sections 5.2.1 to 5.2.4. The significantly increased number of members of such (super)families has rekindled the search for ancestral variants. This research has either led to identification of an ancient prototype such as NrfH in the NapC/NrfH superfamily (see Section 5.1.4) or it has identified gene superfamilies that encode highly substrate-promiscuous enzymes recruited in diverse metabolic modules without answering the question regarding the ancestral type. Phylogenetic reconstructions are sensitive to uneven population of divergent branches with representatives and subject to misrepresentation of extant diversity due to limitations in the discovery process, which largely relies on nucleic acid primers and probes. This calamity was only just recently overcome by significant progress in studies of NC-processing enzymes.

#### 5.2.1. The MCC family

Amongst others, the MCC family comprises the NrfA, Hao and Hzo enzyme families mentioned in previous sections (enzymes 5-8, 15 and 16 in Table 2). Some of its individual members have been investigated repeatedly in the last decade whenever either new functional details or new members were discovered ([38] and references therein). A first comprehensive analysis identified direct evolutionary links between pentaheme (NrfA) and octaheme cytochrome *c* nitrite reductases (Onr and EHao), octaheme tetrathionate reductase (Otr), hyroxylamine (Hao) and hydrazine (Hzo) oxidoreductases and a large number of unidentified octaheme cytochrome c proteins, whose substrate and mechnism of transformation (reduction or oxidation) is still enigmatic, thereby predicting a gene extension model along diversification [52]. This, recent reports on octaheme cytochrome c sulfite reductase [53,162] and numerous new sequences provided reason and opportunity for an extensive new analysis that not only established the new boundaries of a large MCC superfamily (for example by including the tetraheme cytochrome  $c_{554}$  from nitrifiyers) but revised the evolutionary direction along the lines of gene/protein contraction (from octaheme to penta- and tetraheme proteins) and a context for proposing the family's functional origin in the sulfur cycle [53].

#### 5.2.2. The NarGH-NxrAB family

As indicated in Section 3.7, it has been suggested earlier that proteobacterial nitrite oxidoreductase (NxrAB) and nitrate reductases (NarGH) have close genetic relationships and exhibit similar inhibitor profiles [136]. More recent in silico and experimental studies revealed that the process of anaerobic oxidation of nitrite to nitrate in anammox bacteria is facilitated by a Nxr complex (aNxr; enzyme 9c in Table 2), which provides energy and reductant needed for growth [73,74]. A phylogenetic analysis by Lücker et al. [11] demonstrated that the NxrA (homolog of NarG) and NxrB (homolog of NarH) proteins from Nitrobacter species and from Candidatus N. defluvii cluster on opposite branches in the respective subunit protein trees of the type II DMSO reductase superfamily of molybdoproteins. Interestingly, that tree showed that the Nxr proteins from Candidatus N. defluvii cluster with respective proteins from the anammox bacterium Candidatus K. stuttgartiensis in a way that suggests that the enzymes present in extant aerobic Nitrospirae are more delineated over evolutionary time. In a recent article, Sorokin et al. [163] reported the unexpected discovery of an aerobic nitriteoxidizing bacterium in the phylum Chloroflexi, whose NxrAB proteins clustered with those from Nitrobacter and Nitrococcus albeit most closely with the NarGH proteins from the anaerobic methanotroph, Candidatus M. oxyfera in the NC10 phylum. The phylogenetic trees in both studies suggest that nitrite oxidoreductase has evolved at least twice in diverse genomic backgrounds of bacteria adapted to anoxic environments and in reversible metabolic contexts of nitrite oxidation and nitrate reduction. The studies further suggest that the respective genes were likely laterally transferred into genomic backgrounds of different phyla where they assumed key catabolic functions in the chemolithotrophic context of aerobic nitrification.

#### 5.2.3. The Nor/HCO family

Members of the heme copper-oxidase (HCO) superfamily are also known in the literature as terminal cytochrome c oxidases or complex IV oxygen reductases (OR) [164]. Based on extensive comparative genomic, phylogenetic and biochemical analyses, it has been proposed recently that the high-affinity oxygen reductases in the so-called B-, H- and C-families evolved after the low affinity A-family [165]. Numerous papers also established the evolutionary relationship between HCO subunit I and the corresponding subunit of membrane-bound nitric oxide reductases, NorB (see Section 3.3; [166] and references therein). While most authors agree that A-type HCOs are phylogenetically older than the other families, including Nor enzymes, their discussions usually end in speculations when considering that NO is the primordial oxidant and that oxygen at best was sufficiently present locally (maybe as a result of an intra-oxygenic metabolism as it was discovered recently in NC10 bacteria [76,77]) but still at low concentrations before the Great Oxygenation Event (see Section 6). This conundrum is likely the result of oxygen-centric thinking and the assumption that branched electron flow has arisen within a very narrow window of evolutionary time, is constrained to the quinone:cytochrome *c* oxidoreductase of the Rieske  $bc_1$ -type (complex III) and closed by existence of energy conserving complex IV HCOs. In contrast, this review has identified the many different membrane-bound and evolutionarily ancient soluble periplasmic protein complexes that transform nitric oxide by either reduction or oxidation (see Section 3.3). It is thus most parsimonious to predict that early NO reduction was facilitated by soluble, cytochrome *c* or flavocytochrome based inventory that interacted with the quinone/quinol pool via multi cytochrome *c* complexes such as those that are functional as QRP complexes in extant anammox bacteria [73,123] and that NO-mediated electron transfer evolved from circular to linear flow (see also Section 6) and branched off the membrane (quinol pool) without return to a second membrane-bound and potentially energy-conserving module. With this premise, applying the results of phylogenetic analyses (namely that HCO-Nor evolved from HCO-ORs early) and extant metabolic scenarios [165] is without conflict and fits into the overall scheme of that detoxification mechanisms needed to be in place before sophisticated redox (electron flow) systems with high throughput capacity evolved. Hence the absence of Cu<sub>B</sub> centers in Nor enzymes compared to HCO-ORs can be explained as an adaptation by reduction of complexity rather than a reflection of metal availability in different geochemical eras (see Section 6). Interestingly, the high-affinity oxygen reductases in the C-family of HCOs (the cytochrome *cbb*<sub>3</sub> oxidases) have also, albeit a lower, affinity to NO, which has been implicated as an additional NO detoxification function in wastewater-adapted ammonia-oxidizing bacteria [71].

#### 5.2.4. The Cu-MMO family

We learned only recently that members of the copper-containing membrane-bound monooxygenase (Cu-MMO) family are substratepromiscuous enzyme complexes that are encoded in diverse bacterial and archaeal genomic backgrounds [142]. While a common evolutionary history of two of its representatives, Amo and particular methane monooxygenase (pMmo), has been demonstrated for the first time already in 1998 [103] (the frequently cited reference [167] erroneously compared only PmoA sequences and not N. oceani AmoA), it were the very recent findings of Amo in Thaumarchaeaota [88], of pMmo in Verrucomicrobia [168] and NC10 bacteria [76], of ethane-MO (GenBank loci BAH22833 and BAH22839) and ethylene-MO [169] in Proteobacteria, of butane-MO [170] and hexane-MO [171] in Gram-positive bacteria as well as a "pXmo" of unknown substrate specificity in Proteobacteria [142] that led to the correction of a believed narrow taxonomic distribution and dedicated function of Cu-MMOs in bacterial nitrification and methanotrophy. Even though the evolutionary relationship between extant Cu-MMOs has been reconstructed, their origin is still elusive. Their operation is dependent on molecular oxygen and bioavailable copper and all substrates identified so far are reduced oxygen-free nitrogen and carbon compounds. The pMmo operating in strictly anaerobic but intra-oxygenic NC10- bacteria is copper-dependent and not ancestral [142]. Therefore, it is thinkable but yet without experimental

evidence that the Cu-MMO ancestor was actually copper-free and may have utilized NO as the source of oxygen in an environment free of molecular oxygen. Although a thermophilic thaumarchaeal origin of Amo as ancestral Cu-MMO has been hypothesized [90], the fact that Thaumarchaeaota lack cytochrome *c* proteins completely and that their redox protein complement is almost exclusively copper-dependent suggests that their present catabolic machinery is the product of an evolutionarily rather late origin. Additional discoveries will be required to identify genomic backgrounds that have more likely been suited for evolving ancestral Cu-MMO.

### 6. Conclusions on the evolution of bioenergetic systems involved in NC transformation

There is ample evidence supporting the existence of diverse microbial catabolic lifestyles in the anoxic past of planet Earth, many of which including anaerobic respiration [172]. Because of the high frequency of lateral transfer events, it is generally accepted that gains and losses of particular inventory (genes, proteins) or changes in the composition of ancient microbial genomes over evolutionary time is not traceable. It is possible, however, to establish the evolutionary histories of individual inventories found in extant organisms, establish functional associations between them and generate ecophysiological contexts with particular host organisms. Along the same line of thought, it will remain unclear whether primordial enzymes and ETCs that had an impact on nitrogen cycling (and these might have used other metal cofactors than today's enzymes) became extinct during evolution. Another problem arises from the fact that the inventory of NC-converting enzymes is not predictable from the phylogenetic position of a microorganism (usually based on 16S rRNA sequence trees) due to extensive horizontal gene transfer of the respective gene loci. Therefore, evolutionarily "ancient" enzymes are often found in various domains and phyla of extant microbes. In addition, recent discoveries have cemented earlier findings that conversion of particular NCs can be facilitated by multiple and evolutionarily unrelated enzyme complexes; there are, for instance, more than eight different enzyme complexes known to oxidize or reduce NO. Keeping all this in mind, this section discusses conceivable scenarios regarding the evolution of nitrogen cycle bioenergetics that are primarily based on the facts presented in the previous sections of this article, i.e. viewed from the perspective of extant ETC design. In this regard, the following questions are considered relevant:

- (A) Which NCs and enzyme cofactors were present and biologically exploitable in different geochemical eras?
- (B) Which of the presented systems and metabolic pathways are ancient and which are comparatively young on an evolutionary time scale?
- (C) How did the bioenergetic systems in question contribute to the establishment of the global nitrogen cycle as we know it today?

It is conceivable that nitrogen gases such as ammonia, nitric oxide and nitrous oxide were present in the slightly reductive atmosphere of the Archaean era that ended about 2.4 billion years ago, i.e. prior to the Great Oxygenation Event (GOE) that by and large resulted from the emergence of oxygen production by water-splitting phototrophs [173-175]. There is a considerable debate, however, whether nitrogen oxyanions like nitrate and nitrite were available to drive anaerobic respiration before the GOE [143,176]. In principle, nitrate and nitrite might have been formed from nitric oxide under anoxic conditions in the Hadean and Archaean atmosphere [177]. Their concentrations in (micro)environments are difficult to estimate although values as high as in the micromolar range have been suggested [178]. On the other hand, it was argued that nitrate and nitrite only accumulated after their production by aerobic nitrification from ammonium [179], which is known today as an obligatory aerobic process facilitated by copper-dependent catalysis (see below for the bioavailability of copper ions). In this scenario, nitrate and nitrite respiration succeeded the *GOE* [180].

It is widely accepted that copper is unlikely to have played a biological role in the Archaean due to its extremely low solubility (and thus bioavailability) in a sulfidic environment [181]. It has to be assumed that copper only became bioavailable after the GOE when the "sulfidic lock" was abolished [182]. Following this geochemical reasoning, copper enzymes like NirK, NosZ, Amo, HCOs and cupredoxins are seemingly "younger" than those using iron (or porphyrins like heme), molybdenum or nickel as cofactors (Table 2). If true, nitrous oxide reduction and aerobic ammonium oxidation are relatively recent additions to the metabolic repertoire of NC-converting organisms. However, there might have been functionally similar but copper-independent enzymes that probably were abandoned when the biosynthetically less costly copper-dependent enzymes appeared. For instance, the function of the diiron cluster in Cu-MMOs in context with the reactive Cu<sub>B</sub> sites is still not resolved [104,105] and leaves room for speculations of an "iron first, copper second" scenario in the evolution of this large enzyme superfamily. Consequently and according to metal availability predictions, NarGH, NapA, cytochromes *b* and *c* (including cytochrome *bc* complexes and MCCs) and all QRPs shown in Table 4 might have been formed in the pre-GOE era. Therefore, anaerobic processes like nitrate reduction, nitrite reduction to nitric oxide, nitrous oxide or ammonium and the anammox pathway could have been operational in the Archaean, and the only reasonable process to close the nitrogen cycle and return fixed nitrogen to the dinitrogen pool was the anammox process [56]. A few more hypotheses can be put forward with respect to assumed properties of "older" enzymes: (i) exported enzymes [requiring protein export by the ancient twin arginine translocation (Tat) system] are older than those that convert NCs in the cytoplasm, thus requiring substrate/product transporters (especially for toxic products like nitrite in the case of nNar of *c*Nar); (ii) enzymes with activities playing a role in the sulfur cycle (like PsrC and members of the MCC and Nap/NrfH families) are early inventions [183]; (iii) enzymes reacting directly with redox mediator pools (quinone/quinol and periplasmic cytochrome *c* pools) are older than those forming tight complexes with QRPs or cytochromes [56]; (iv) enzymes operational at low substrate concentrations are older, (v) enzymes that resulted from the fusion of otherwise separated protein domains are younger and their genes more likely prone to horizontal transfer; (vi) within enzyme families, proteins showing special active sites or requiring dedicated biogenesis systems (like some members of the MCC family) are younger inventions [184-191].

Early bioenergetic systems likely have evolved from detoxification systems that inactivated reactive nitrogen species (nitric oxide and nitrite) and possibly also toxic sulfur compounds like sulfite in the periplasm and outside the cell. In this respect, it is noteworthy that the  $d_1$  heme of cytochrome  $cd_1$  nitrite reductase is synthesized from siroheme, the cofactor of several cytoplasmic enzymes such as respiratory sulfite reductase and assimilatory nitrite reductase [192]. Moreover, nitric oxide transforming enzymes are present in Archaea and many different bacterial phyla and constitute an unprecedented high abundance and diversity [4,114]. According to the assumptions made above, the Nar, NirS and Nrf-type modules also qualify as ancient enzyme systems. Thus, denitrification of nitrate to yield nitrous oxide as well as nitrate ammonification appears to constitute significant processes in the primordial unclosed nitrogen cycle [56]. These pathways might have been electroneutral in ancestral microbes, using available high-potential NCs as electron sinks in order to replenish the quinone pool and employing ETCs that consisted of simple enzymes modules depending on hemes and iron-sulfur centers. It can thus be predicted that small cytochromes c and iron-sulfur proteins such as ferredoxins have been the principle redox mediators in the extracellular space and that the cytochrome  $bc_1$  complex as well as other QRPs like the NapH/ NosH, NrfD/PsrC and NapC/NrfH families represent the complement of early respiratory modules.

It cannot not be excluded that nitric oxide dismutase is also an ancient enzyme that may have made oxygen bioavailable intracellularly for respiratory processes, a situation that likely would not left any biogeochemical isotope record. Such an oxygen source might have provided significant amounts of oxygen locally prior to the GOE that could have served as substrate for electrogenic, low-affinity HCOs in the A-family (see Section 5.2.3) [165,179,193]. In contrast to the nitric oxide reductases that evolved from the high affinity oxygen reductases in the HCO B- and H-families, extant representatives of the two sister lineages of nitric oxide reductases, cNor and qNor (which evolved from the HCO C-family of cytochrome *cbb*<sub>3</sub> high-affinity oxygen reductases), contain several non-heme irons and no copper in their active sites, which has provided the impetus for studying unusual representatives of the qNor family as candidates of the nitric oxide dismutase (see Section 3.3). For example, the genome of Candidatus M. oxyfera encodes one qNor and two versions of qNor-like proteins and sequence-similar variants have been detected in other bacteria; however, to date there is no physiological evidence that they perform that reaction (James Hemp, personal communication). In this context and provided that proper detoxification systems were in place, it is thinkable that microaerobic low-throughput oxidation of reduced carbon and nitrogen compounds by soluble monooxygenases such as the soluble iron-dependent methane monooxygenase (that still operates in a number of extant methanotrophs) or a primordial copper-independent substratepromiscuous membrane-bound monooxygenase constituted the reductive branches of early supplementary catabolism in a largely anoxic environment [114,142]. The evolution of such an enzymic inventory occurred likely in the genomes of sulfur-dependent anaerobic bacteria [56] and an association with microbes with intra-oxygenic capacity may have locally increased the bioavailability of sulfur-trapped copper [181]. If true, the evolution of copper-associated membrane-bound monooxygenases (Cu-MMO enzymes) and the recruitment of their hosts into nitrification aggregates by Nxr-containing bacteria could have led to the production of nitrite and nitrate prior to the GOE, contrary to what was suggested by others [167,180,194] before the recent discovery of anaerobic nitrite oxidation to nitrate by anammox (see below)

It has been proposed that representatives of the MCC family have participated in early NC transformations before there was a closed nitrogen cycle [56], which occurred mainly in two capacities: the reduction of nitrite, NO and hydroxylamine (NrfA, Onr, cytochrome  $c_{554}$ ,  $\epsilon$ Hao) to ammonium and the oxidation of hydroxylamine (Hao) to NO or nitrite [32,52,53]. The other main complement of early NC transformations were the reductions of nitrate to nitrite and to nitric oxide by the molybdoproteins NapA and NarG, by NirS (cyt  $cd_1$ , which has also low affinity for oxygen [195]) as well as by soluble tetraheme cytochromes c' (CytS, CytP). All of these modules likely evolved as means for detoxification, some of which with utility in anaerobic respiration. The major breakthrough in terms of employing these reactions for a productive catabolic lifestyle likely occurred in anammox bacteria that were able to functionally unite these modules by invention of a cytochrome c protein complex capable of comproportionating NO and ammonium to hydrazine (hydrazine synthase) and acquisition of NirS [74] (note that some anammox bacteria contain a NirK homolog [196]). This was possible because MCCs were capable of detoxifying hydrazine by oxidation to dinitrogen, which not only provided reductant for *pmf* generation but also established the first (closed) nitrogen cycle [56,137]. Nxr (Type II DMSO reductase family) was employed to oxidize nitrite to nitrate anaerobically [11,163], which provided anammox bacteria with the needed energy for assimilation and biosynthesis [123] and was likely the main biological way of nitrate production before aerobic nitrification emerged. Once oxygen concentrations rose past threshold levels that led to the emergence of copper-containing enzymes (NirK, cupredoxins with electron storage function, NosZ and Amo), some of the anammox modules were replaced by aerobic, high-throughput modules (i.e., soluble periplasmic NirS and hydrazine synthase were replaced with a Cu-MMO), which provided metabolic control and more efficient use of reductant thereby enabling linear electron flow and branched ETCs for better energy conservation [56]. Despite this adaptation and reliance on molecular oxygen as the main oxidant in oxic and hypoxic environments, both anaerobic and aerobic microbes with active roles in the extant nitrogen cycle have retained and express an arsenal of NOtransforming modules. It can therefore be stated that the reactive nitrogen detoxification mechanisms shown in Table 1 constituted not only the prerequisites for the evolution of nitrogen-based catabolic pathways but must continue to play important roles despite the presence of oxygen as the most powerful oxidant.

#### 7. Perspective

One of the most astonishing features of the presented microbial ETCs is their high degree of modularity that appears to be a major prerequisite for optimized electron transport to and from available NCs. Interestingly, it turned out that the various dehydrogenase and reductase complexes are electronically connected by only a few types of soluble electron transfer proteins (mainly small c-type cytochromes and cupredoxins) and by a limited number of QRP modules. Furthermore, *pmf* generation seems to depend on the functional architecture of only a few of these modules while most of them merely serve in efficient electron transport and substrate turnover. Genomic studies clearly revealed that genes encoding NC-converting enzymes as well as their corresponding electron transport modules have been arranged in many different contexts during evolution and it is quite obvious that these genetic arrangements are well suited for the acquisition of new metabolic features by horizontal gene transfer (despite the fact that elaborate biogenesis systems are needed to achieve the maturation of some of the discussed enzyme systems). On the other hand, extensive module tinkering makes it very difficult to reconstruct the evolution of bioenergetic systems (rather than that of individual enzymes) since the Archaean era. Despite all uncertainties, this article outlined some possible scenarios that are in line with biochemical and well as environmental and geological contraints and concepts [180,197,198].

Unfortunately, many electron transfer reactions discussed in this article have not been demonstrated biochemically (for example using purified enzymes or by reconstitution experiments in case of ORP-quinone interaction) but were rather predicted from the individual characterization of potentially interacting enzymes or from the composition of corresponding gene clusters. Especially in light of the emerging wealth of genomic data, it is critical that modelling electron transfer chains or networks is not exclusively based on genetic information alone (despite the fact that many models established in this way proved to be correct later on). One also has to realize that only a very small proportion of the known (and cultured) bacteria have been investigated in depth and surely we can await many future surprises in terms of the biochemical characterization of ETCs involved in nitrogen metabolism. Furthermore, crystal structures of many key enzymes discussed in this article are still lacking, for example those of Amo, Nxr, EHao, Hzs, NapH or NapC.

One of the most discussed challenges in modern microbiology is the rational design of synthetic microorganisms that can be applied to preserve or restore human and/or environmental health. Regarding the nitrogen cycle, this is evident when considering global warming and the increases in atmospheric nitrous oxide and methane concentrations due to significant anthropogenic overloads of fixed nitrogen (i.e. Haber-Bosch process produced ammonium as fertilizer) to the nitrogen cycle [79,180,199,200]. Synthesizing new or changing existing organisms clearly also requires proper design of energy metabolism, for which more and in-depth knowledge of substrate conversion mechanisms and enzyme interactions that operate in extant microorganisms is necessary (not to mention the underlying sensing mechanisms and regulatory pathways). Furthermore, a more detailed knowledge on enzyme maturation processes, synthesis of co-factors and their insertion into proteins as well as of transport processes of apo- or holoproteins across membranes is required. Progress in this field also depends on efficient heterologous production systems for metalloenzymes or enzyme complexes in suitable host bacteria. Ultimately, the combined knowledge may lead to the *de novo* construction of organisms whose energy metabolism meets favored requirements in industrial and environmental applications.

#### Acknowledgements

The authors thank their co-workers of cited publications for contributions to various aspects of microbial physiology and the architecture of respiratory electron transport chains. MGK is funded by the US-NSF (grants EF-0541797 and MCD-1202648) and incentive funds provided by the University of North Carolina. Work in the laboratory of JS is funded by the Deutsche Forschungsgemeinschaft (grants SI 848/4-1 and SI 848/5-1).

#### References

- In: B.B. Ward, D.J. Arp, M.G. Klotz (Eds.), Nitrification, ASM Press, Washington D.C., 2011.
- [2] In: J.W.B. Moir (Ed.), Nitrogen Cycling in Bacteria, Molecular Analysis, Caister Academic Press, Norfolk, UK, 2011.
- [3] In: H. Bothe, S.J. Ferguson, W.E. Newton (Eds.), Biology of the Nitrogen Cycle, Elsevier, Amsterdam, The Netherlands, 2007.
- [4] L.Y. Stein, M.G. Klotz, Nitrifying and denitrifying pathways of methanotrophic bacteria, Biochem. Soc. Trans. 39 (2011) 1826–1831.
- [5] D.G. Nicholls, S.J. Ferguson, Bioenergetics 3, Academic Press, Amsterdam, 2002.
- [6] J. Simon, R.J.M. van Spanning, D.J. Richardson, The organisation of proton motive and non-proton motive redox loops in prokaryotic respiratory systems, Biochim. Biophys. Acta 1777 (2008) 1480–1490.
- [7] D. Richardson, Redox complexes of the nitrogen cycle, in: J.W.B. Moir (Ed.), Nitrogen Cycling in Bacteria, Caister Academic Press, Norfolk, UK, Molecular Analysis, 2011, pp. 23–37.
- [8] J. Simon, Organization of respiratory electron transport chains in nitrate-reducing and nitrifying bacteria, in: J.W.B. Moir (Ed.), Nitrogen Cycling in Bacteria, Molecular Analysis, Caister Academic Press, Norfolk, UK, 2011, pp. 39–58.
- [9] B. Kraft, M. Strous, H.E. Tegetmeyer, Microbial nitrate respiration-genes, enzymes and environmental distribution, J. Biotech. 155 (2011) 104-117.
- [10] J.F. Stolz, P. Basu, Evolution of nitrate reductase: molecular and structural variations on a common function, Chembiochem 3 (2002) 198–206.
- [11] S. Lücker, M. Wagner, F. Maixner, E. Pelletier, H. Koch, B. Vacherie, T. Rattei, J.S.S. Damsté, E. Spieck, D. Le Paslier, H. Daims, A *Nitrospira* metagenome illuminates the physiology and evolution of globally important nitrite-oxidizing bacteria, Proc. Natl. Acad. Sci. U. S. A. 107 (2010) 13479–13484.
- [12] A. Magalon, J.G. Fedor, A. Walburger, J.H. Weiner, Molybdenum enzymes in bacteria and their maturation, Coord. Chem. Rev. 255 (2011) 1159–1178.
- [13] D. Richardson, G. Sawers, PMF through the redox loop, Science 295 (2002) 1842–1843.
- [14] W. Jia, J.A. Cole, Nitrate and nitrite transport in *Escherichia coli*, Biochem. Soc. Trans. 33 (2005) 159–161.
- [15] J.W.B. Moir, N.J. Wood, Nitrate and nitrite transport in bacteria, Cell. Mol. Life Sci. 58 (2001) 215–224.
- [16] B. Ize, S.J. Coulthurst, K. Hatzixanthis, I. Caldelari, G. Buchanan, E.C. Barclay, D.J. Richardson, T. Palmer, F. Sargent, Remnant signal peptides on non-exported enzymes: implications for the evolution of prokaryotic respiratory chains, Microbiology 155 (2009) 3992–4004.
- [17] F. Cava, O. Zafra, J. Berenguer, A cytochrome c containing nitrate reductase plays a role in electron transport for denitrification in *Thermus thermophilus* without involvement of the respiratory bc respiratory complex, Mol. Microbiol. 70 (2008) 507–518.
- [18] P. Cabello, M.D. Roldán, C. Moreno-Vivián, Nitrate reduction and the nitrogen cycle in Archaea, Microbiology 150 (2004) 3527–3546.
- [19] R.M. Martinez-Espinosa, E.J. Bridge, M.J. Bonete, J.N. Butt, C.S. Butler, F. Sargent, D.J. Richardson, Look on the positive side! The orientation, identification and bioenergetics of "archaeal" membrane-bound nitrate reductases, FEMS Microbiol. Lett. 276 (2007) 129–139.
- [20] S. de Vries, M. Momcilovic, M.J.F. Strampraad, J.P. Whitelegge, A. Baghai, I. Schröder, Adaptation to a high-tungsten environment: *Pyrobaculum aerophilum* contains an active tungsten nitrate reductase, Biochemistry 49 (2010) 9911–9921.
- [21] D.J. Richardson, Bacterial respiration: a flexible process for a changing environment, Microbiology 146 (2000) 551–571.
- [22] D.J. Richardson, B.C. Berks, D.A. Russell, S. Spiro, C.J. Taylor, Functional, biochemical and genetic diversity of prokaryotic nitrate reductases, Cell. Mol. Life Sci. 58 (2001) 165–178.
- [23] S. Grahl, J. Maillard, C.A.E.M. Spronk, G.W. Vuister, F. Sargent, Overlapping transport and chaperone-binding functions within a bacterial twin-arginine signal peptide, Mol. Microbiol. 83 (2012) 1254–1267.
- [24] A. Nilavongse, T.H.C. Brondijk, T.W. Overton, D.J. Richardson, E.R. Leach, J.A. Cole, The NapF protein of the *Escherichia coli* periplasmic nitrate reductase system:

demonstration of a cytoplasmic location and interaction with the catalytic subunit, NapA, Microbiology 152 (2006) 3227–3237.

- [25] M.F. Olmo-Mira, M. Gavira, D.J. Richardson, F. Castillo, C. Moreno-Vivián, M.D. Roldán, NapF is a cytoplasmic iron-sulfur protein required for Fe S cluster assembly in the periplasmic nitrate reductase, J. Biol. Chem. 279 (2004) 49727–49735.
- [26] M. Kern, A. Mager, J. Simon, Role of individual nap gene cluster products in NapC-independent nitrate respiration of Wolinella succinogenes, Microbiology 153 (2007) 3739–3747.
- [27] M. Kern, J. Simon, Periplasmic nitrate reduction in Wolinella succinogenes: cytoplasmic NapF facilitates NapA maturation and requires the menaquinol dehydrogenase NapH for membrane attachment, Microbiology 155 (2009) 2784–2794.
- [28] T.H.C. Brondijk, D. Fiegen, D.J. Richardson, J.A. Cole, Roles of NapF, NapG and NapH, subunits of the *Escherichia coli* periplasmic nitrate reductase, in ubiquinol oxidation, Mol. Microbiol. 44 (2002) 245–255.
- [29] J. Simon, M. Sänger, S.C. Schuster, R. Gross, Electron transport to periplasmic nitrate reductase (NapA) of *Wolinella succinogenes* is independent of a NapC protein, Mol. Microbiol. 49 (2003) 69–79.
- [30] S.M. Sievert, K.M. Scott, M.G. Klotz, P.S.G. Chain, LJ. Hauser, J. Hemp, M. Hügler, M. Land, A. Lapidus, F.W. Larimer, S. Lucas, S.A. Malfatti, F. Meyer, I.T. Paulsen, Q. Ren, J. Simon, USF Genomics Class, Genome of the epsilonproteobacterial chemolithoautotroph *Sulfurimonas denitrificans*, Appl. Environ. Microbiol. 74 (2008) 1145–1156.
- [31] H. Gao, Z.K. Yang, S. Barua, S.B. Reed, M.F. Romine, K.H. Nealson, J.K. Fredrickson, J.M. Tiedje, J. Zhou, Reduction of nitrate in *Shewanella oneidensis* depends on atypical NAP and NRF systems with NapB as a preferred electron transport protein from CymA to NapA, ISME J. 3 (2009) 966–976.
- [32] B.J. Campbell, J.L. Smith, T.E. Hanson, M.G. Klotz, L.Y. Stein, C.K. Lee, D. Wu, J.M. Robinson, H.M. Khouri, J.A. Eisen, S.C. Cary, Adaptations to submarine hydrothermal environments exemplified by the genome of *Nautilia profundicola*, PLoS Genet. 5 (2) (2009) e1000362.
- [33] S.-H. Kim, C. Harzman, J.K. Davis, R. Hutcheson, J.B. Broderick, T.L. Marsh, J.A. Tiedje, Genome sequence of *Desulfitobacterium hafniense* DCB-2, a Gram-positive anaerobe capable of dehalogenation and metal reduction, BMC Microbiol. 12 (2012) 21.
- [34] I.V. Pearson, M.D. Page, R.J.M. van Spanning, S.J. Ferguson, A mutant of *Paracoccus denitrificans* with disrupted genes coding for cytochrome c<sub>550</sub> and pseudoazurin establishes these two proteins as the in vivo electron donors to cytochrome cd<sub>1</sub> nitrite reductase, J. Bacteriol. 185 (2003) 6308–6315.
- [35] M. Nojiri, H. Koteishi, T. Nakagami, K. Kobayashi, T. Inoue, K. Yamaguchi, S. Suzuki, Structural basis of inter-protein electron transfer for nitrite reduction in denitrification, Nature 462 (2009) 117–120.
- [36] A. Jüngst, S. Wakabayashi, H. Matsubara, W.G. Zumft, The *nirSTBM* region coding for cytochrome *cd*<sub>1</sub>-dependent nitrite respiration of *Pseudomonas stutzeri* consists of a cluster of mono-, di-, and tetraheme proteins, FEBS Lett. 279 (1991) 205–209.
- [37] M.J. Ellis, J.G. Grossmann, R.R. Eady, S.S. Hasnain, Genomic analysis reveals widespread occurrence of new classes of copper nitrite reductases, J. Biol. Inorg. Chem. 12 (2007) 1119–1127.
- [38] J. Simon, M. Kern, B. Hermann, O. Einsle, J.N. Butt, Physiological function and catalytic versatility of bacterial multihaem cytochromes *c* involved in nitrogen and sulfur cycling, Biochem. Soc. Trans. 39 (2011) 1864–1870.
- [39] J. Simon, R. Gross, O. Einsle, P.M.H. Kroneck, A. Kröger, O. Klimmek, A NapC/NirT-type cytochrome c (NrfH) is the mediator between the quinone pool and the cytochrome c nitrite reductase of Wolinella succinogenes, Mol. Microbiol. 35 (2000) 686–696.
- [40] M.L. Rodrigues, T.F. Oliveira, I.A.C. Pereira, M. Archer, X-ray structure of the membrane-bound cytochrome *c* quinol dehydrogenase NrfH reveals novel haem coordination, EMBO J. 25 (2006) 5951–5960.
- [41] M. Kern, O. Einsle, J. Simon, Variants of the tetrahaem cytochrome *c* quinol dehydrogenase NrfH characterize the menaquinol binding site, the haem *c* binding motifs and the transmembrane segment, Biochem. J. 414 (2008) 73–79.
- [42] J. Simon, Enzymology and bioenergetics of respiratory nitrite ammonification, FEMS Microbiol. Rev. 26 (2002) 285–309.
- [43] S.R. Poock, E.R. Leach, J.W.B. Moir, J.A. Cole, D.J. Richardson, Respiratory detoxification of nitric oxide by the cytochrome *c* nitrite reductase of *Escherichia coli*, J. Biol. Chem. 277 (2002) 23664–23669.
- [44] R.K. Poole, Nitric oxide and nitrosative stress tolerance in bacteria, Biochem. Soc. Trans. 33 (2005) 176–180.
- [45] P.C. Mills, G. Rowley, S. Spiro, J.C.D. Hinton, D.J. Richardson, A combination of cytochrome c nitrite reductase (NrfA) and flavorubredoxin (NorV) protects Salmonella enterica serovar Typhimurium against killing by NO in anoxic environments, Microbiology 154 (2008) 1218–1228.
- [46] M. Kern, J. Volz, J. Simon, The oxidative and nitrosative stress defence network of *Wolinella succinogenes*: cytochrome c nitrite reductase mediates the stress response to nitrite, nitric oxide, hydroxylamine and hydrogen peroxide, Environ. Microbiol. 13 (2011) 2478–2494.
- [47] M. Jormakka, K. Yokoyama, T. Yano, M. Tamakoshi, S. Akimoto, T. Shimamura, P. Curmi, S. Iwata, Molecular mechanism of energy conservation in polysulfide respiration, Nat. Struct. Mol. Biol. 15 (2008) 730–737.
- [48] J. Simon, M. Kern, Quinone-reactive proteins devoid of haem b form widespread membrane-bound electron transport modules in bacterial anaerobic respiration, Biochem. Soc. Trans. 36 (2008) 1011–1016.
- [49] T.A. Clarke, J.A. Cole, D.J. Richardson, A.M. Hemmings, The crystal structure of the pentahaem c-type cytochrome NrfB and characterization of its solution-state interaction with the pentahaem nitrite reductase NrfA, Biochem. J. 406 (2007) 19–30.
- [50] H.J. Kim, A. Zatsman, A.K. Upadhyay, M. Whittaker, D. Bergmann, M.K. Hendrich, A.B. Hooper, Membrane tetraheme cytochrome c<sub>m552</sub> of the ammonia-oxidizing *Nitrosomonas europaea*: a ubiquinone reductase, Biochemistry 47 (2008) 6539–6551.

- [51] D.J. Bergmann, A.B. Hooper, M.G. Klotz, Structure and sequence conservation of genes in the *hao* cluster of autotrophic ammonia-oxidizing bacteria: evidence for their evolutionary history, Appl. Environ. Microbiol. 71 (2005) 5371–5382.
- [52] M.G. Klotz, M.C. Schmid, M. Strous, H.J.M. op den Camp, M.S.M. Jetten, A.B. Hooper, Evolution of an octahaem cytochrome *c* protein family that is key to aerobic and anaerobic ammonia oxidation by bacteria, Environ. Microbiol. 10 (2008) 3150–3158.
- [53] M. Kern, M.G. Klotz, J. Simon, The Wolinella succinogenes mcc gene cluster encodes an unconventional respiratory sulfite reduction system, Mol. Microbiol. 82 (2011) 1515–1530.
- [54] M. Kern, J. Simon, Electron transport chains and bioenergetics of respiratory nitrogen metabolism in *Wolinella succinogenes* and other Epsilonproteobacteria, Biochim. Biophys. Acta 1787 (2009) 646–656.
- [55] D.J. Arp, P.S.G. Chain, M.G. Klotz, The impact of genome analyses on our understanding of ammonia-oxidizing bacteria, Annu. Rev. Microbiol. 61 (2007) 21–58.
- [56] M.G. Klotz, LY. Stein, Nitrifier genomics and evolution of the nitrogen cycle, FEMS Microbiol. Lett. 278 (2008) 146–156.
   [57] B.O. Flmore, D.I. Bergmann, M.G. Klotz, A.B. Hooper, Cytochromes P460 and c'-beta.
- [57] B.O. Elmore, D.J. Bergmann, M.G. Klotz, A.B. Hooper, Cytochromes P460 and c'-beta; a new family of high-spin cytochromes c, FEBS Lett. 581 (2007) 911–916.
- [58] R. Cross, D. Lloyd, R.K. Poole, J.W.B. Moir, Enzymatic removal of nitric oxide catalyzed by cytochrome c' in *Rhodobacter capsulatus*, J. Bacteriol. 183 (2001) 3050–3054.
- [59] M. Deeudom, M. Koomey, J.W.B. Moir, Roles of *c*-type cytochromes in respiration in *Neisseria meningitidis*, Microbiology 154 (2008) 2857–2864.
- [60] N.J. Gilberthorpe, R.K. Poole, Nitric oxide homeostasis in Salmonella typhimurium, J. Biol. Chem. 283 (2008) 11146–11154.
- [61] A.K. Upadhyay, A.B. Hooper, M.P. Hendrich, NO reductase activity of the tetraheme cytochrome c<sub>554</sub> of *Nitrosomonas europaea*, J. Am. Chem. Soc. 128 (2006) 4330–4337.
- [62] T.M. Stevanin, J.W.B. Moir, R.C. Read, Nitric oxide detoxification systems enhance survival of *Neisseria meningitidis* in human macrophages and in nasopharyngeal mucosa, Infect. Immun. 73 (2005) 3322–3329.
- [63] C. Costa, A. Macedo, I. Moura, J.J.G. Moura, J. LeGall, Y. Berlier, M.-Y. Liu, W.J. Payne, Regulation of the hexaheme nitrite/nitric oxide reductase of *Desulfovibrio desulfuricans*, *Wolinella succinogenes* and *Escherichia coli*, FEBS Lett. 276 (1990) 67–70.
- [64] J. Hendriks, A. Oubrie, J. Castresana, A. Urbani, S. Gemeinhardt, M. Saraste, Nitric oxide reductases in bacteria, Biochim. Biophys. Acta 1459 (2000) 266–273.
- [65] W.G. Zumft, Nitric oxide reductases of prokaryotes with emphasis on the respiratory, heme-copper oxidase type, J. Inorg. Chem. 99 (2005) 194–215.
- [66] J. Hemp, R.B. Gennis, Diversity of the heme-copper superfamily in Archaea: insights from genomics and structural modeling, Results Probl. Cell Differ. 45 (2008) 1–31.
- [67] T. Hino, Y. Matsumoto, S. Nagano, H. Sugimoto, Y. Fukumori, T. Murata, S. Iwata, Y. Shiro, Structural basis of biological N<sub>2</sub>O generation by bacterial nitric oxide reductase, Science 330 (2010) 1666–1670.
- [68] Y. Matsumoto, T. Tosha, A.V. Pisliakov, T. Hino, H. Sugimoto, S. Nagano, Y. Sugita, Y. Shiro, Crystal structure of quinol-dependent nitric oxide reductase from *Geobacillus stearothermophilus*, Nat. Struct. Mol. Biol. 19 (2012) 238–245.
- [69] J. Hendriks, A. Warne, U. Gohlke, T. Haltia, C. Ludovici, M. Lübben, M. Saraste, The active site of the bacterial nitric oxide reductase is a dinuclear iron center, Biochemistry 37 (1998) 13102–13109.
- [70] S. de Vries, L. Suharti, A.M. Pouvreau, Nitric oxide reductase: structural variations and catalytic mechanism, in: H. Bothe, S.J. Ferguson, W.E. Newton (Eds.), Biology of the Nitrogen Cycle, Elsevier, Amsterdam, The Netherlands, 2007, pp. 57–66.
- [71] L.Y. Stein, D.J. Arp, P.M. Berube, P.S.G. Chain, LJ. Hauser, M.S.M. Jetten, M.G. Klotz, F.W. Larimer, J.M. Norton, H.J.M. Op den Camp, M. Shin, X. Wei, Whole-genome analysis of the ammonia-oxidizing bacterium, *Nitrosomonas eutropha* C91: implications for niche adaptation, Environ. Microbiol. 9 (2007) 1–15.
- [72] Y.-W. Lin, N. Yeung, Y.-G. Gao, K.D. Miner, L. Lei, H. Robinson, Y. Lu, Introducing a 2-His-1-Glu nonheme iron center into myoglobin confers nitric oxide reductase activity, J. Am. Chem. Soc. 132 (2010) 9970–9972.
- [73] B. Kartal, W.J. Maalcke, N.M. de Almeida, I. Cirpus, J. Gloerich, W. Geerts, H.J.M. Op den Camp, H.R. Harhangi, E.M. Janssen-Megens, K.-J. Francoijs, H.G. Stunnenberg, J.T. Keltjens, M.S.M. Jetten, M. Strous, Molecular mechanism of anaerobic ammonium oxidation, Nature 479 (2011) 127–130.
- [74] J. van de Vossenberg, D. Woebken, W.J. Maalcke, H.J.C.T. Wessels, B.E. Dutilh, B. Kartal, E.M. Janssen-Megens, G. Roeselers, J. Yan, D. Speth, J. Gloerich, W. Geerts, E. van der Biezen, W. Pluk, K.-J. Francoijs, L. Russ, P. Lam, S.A. Malfatti, S. Green Tringe, S.C.M. Haaijer, H.J.M. Op den Camp, H.G. Stunnenberg, R. Amann, M.M.M. Kuypers, M.S.M. Jetten, The metagenome of the marine anammox bacterium *Scalindua profunda* illustrates the versatility of this globally important nitrogen cycle bacterium, Environ. Microbiol. 14 (2012), http://dx.doi.org/10.1111/j.1462-2920.2012.02774.x (Early view).
- [75] A.A. Raghoebarsing, A. Pol, K.T. van de Pas-Schoonen, A.J.P. Smolders, K.F. Ettwig, W.I.C. Rijpstra, S. Schouten, H.J.M. Op den Camp, M.S.M. Jetten, M. Strous, A microbial consortium couples anaerobic methane oxidation to denitrification, Nature 440 (2006) 918–921.
- [76] K.F. Ettwig, M.K. Butler, D. Le Paslier, E. Pelletier, S. Mangenot, M.M.M. Kuypers, F. Schreiber, B.E. Dutilh, J. Zedelius, D. de Beer, J. Gloerich, H.J.C.T. Wessels, T. van Alen, F. Luesken, M.L. Wu, K.T. van de Pas-Schoonen, H.J.M. Op den Camp, E.M. Janssen-Megens, K.-J. Francoijs, H. Stunnenberg, J. Weissenbach, M.S.M. Jetten, M. Strous, Nitrite-driven anaerobic methane oxidation by oxygenic bacteria, Nature 464 (2010) 543–548.
- [77] M.L. Wu, K.F. Ettwig, M.S.M. Jetten, M. Strous, J.T. Keltjens, L. van Niftrik, A new intra-aerobic metabolism in the nitrite-dependent anaerobic methane-oxidizing

bacterium Candidatus 'Methylomirabilis oxyfera', Biochem. Soc. Trans. 39 (2011) 243–248.

- [78] K.F. Ettwig, T. van Alen, K.T. van de Pas-Schoonen, M.S.M. Jetten, M. Strous, Enrichment and molecular detection of denitrifying methanotrophic bacteria of the NC10 phylum, Appl. Environ. Microbiol. 75 (2009) 3656–3662.
- [79] D. Richardson, H. Felgate, N. Watmough, A. Thomson, E. Baggs, Mitigating release of the potent greenhouse gas N<sub>2</sub>O from the nitrogen cycle–could enzymic regulation hold the key? Trends Biotechnol. 27 (2009) 388–397.
- [80] W.G. Zumft, Biogenesis of the bacterial respiratory Cu<sub>A</sub>, Cu S enzyme nitrous oxide reductase, J. Mol. Microbiol. Biotechnol. 10 (2005) 154–166.
- [81] W.G. Zumft, P.M.H. Kroneck, Respiratory transformation of nitrous oxide (N<sub>2</sub>O) to dinitrogen by *Bacteria* and *Archaea*, Adv. Microb. Physiol. 52 (2007) 107–227.
   [82] A. Pomowski, W.G. Zumft, P.M.H. Kroneck, O. Einsle, N<sub>2</sub>O binding at a [4Cu:2S]
- copper-sulphur cluster in nitrous oxide reductase, Nature 477 (2011) 234–237. [83] S. Teraguchi, T.C. Hollocher, Purification and some characteristics of a cytochrome
- c-containing nitrous oxide redcutase from Wolinella succinogenes, J. Biol. Chem. 264 (1989) 1972–1979.
- [84] J. Simon, O. Einsle, P.M.H. Kroneck, W.G. Zumft, The unprecedented nos gene cluster of Wolinella succinogenes encodes a novel respiratory electron transfer pathway to cytochrome c nitrous oxide reductase, FEBS Lett. 569 (2004) 7–12.
- [85] P. Wunsch, W.G. Zumft, Functional domains of NosR, a novel transmembrane iron-sulfur flavoprotein necessary for nitrous oxide respiration, J. Bacteriol. 187 (2005) 1992–2001.
- [86] D.J. Arp, L.A. Sayavedra-Soto, N.G. Hommes, Molecular biology and biochemistry of ammonia oxidation by *Nitrosomonas europaea*, Arch. Microbiol. 178 (2002) 250–255.
- [87] A.B. Hooper, T. Vannelli, D.J. Bergmann, D.M. Arciero, Enzymology of the oxidation of ammonia to nitrite by bacteria, Antonie Van Leeuwenhoek 71 (1997) 59–67.
- [88] M. Könneke, A.E. Bernhard, J.R. de la Torre, C.B. Walker, J.B. Waterbury, D.A. Stahl, Isolation of an autotrophic ammonia-oxidizing marine archaeon, Nature 437 (2005) 543–546.
- [89] A.B. Hooper, D.M. Arciero, D. Bergmann, M.P. Hendrich, The oxidation of ammonia as an energy source in bacteria in respiration, in: D. Zannoni (Ed.), Respiration in Archaea and Bacteria: Diversity of Procaryotic Respiratory Systems, Springer, Dordrecht, The Netherlands, 2005, pp. 121–147.
- [90] C.B. Walker, J.R. de la Torre, M.G. Klotz, H. Urakawa, N. Pinel, D.J. Arp, C. Brochier-Armanet, P.S. Chain, P.P. Chan, A. Gollabgir, J. Hemp, M. Hügler, E.A. Karr, M. Könneke, M. Shin, T.J. Lawton, T. Lowe, W. Martens-Habbena, L.A. Sayavedra-Soto, D. Lang, S.M. Sievert, A.C. Rosenzweig, G. Manning, D.A. Stahl, *Nitrosopumilus maritimus* genome reveals unique mechanisms for nitrification and autotrophy in globally distributed marine crenarchaea, Proc. Natl. Acad. Sci. U, S. A. 107 (2010) 8818–8823.
- [91] C. Schleper, G.W. Nicol, Ammonia oxidizing Archaea-genomes, physiology and ecology, Adv. Microb. Physiol. 57 (2010) 1–41.
- [92] J.J.L. Cantera, L.Y. Stein, Role of nitrite reductase in the ammonia-oxidizing pathway of *Nitrosomonas europaea*, Arch. Microbiol. 188 (2007) 349–354.
- [93] R. Bartossek, G.W. Nicol, A. Lanzen, H.-P. Klenk, C. Schleper, Homologues of nitrite reductases in ammonia-oxidizing Archaea: diversity and genomic context, Environ. Microbiol. 12 (2010) 1075–1088.
- [94] J.T. Hollibaugh, S. Gifford, S. Sharma, N. Bano, M. Moran, Metatranscriptomic analysis of ammonia-oxidizing organisms in an estuarine bacterioplankton assemblage, ISME J. 5 (2011) 866–878.
- [95] J.J. Alzerreca, J.M. Norton, M.G. Klotz, The amo operon in marine, ammonia-oxidizing Gammaproteobacteria, FEMS Microbiol. Lett. 180 (1999) 21–29.
- [96] M.G. Klotz, J. Alzerreca, J.M. Norton, A gene encoding a membrane protein exists upstream of the *amoA/amoB* genes in ammonia oxidizing bacteria: a third member of the *amo* operon? FEMS Microbiol. Lett. 150 (1997) 65–73.
- [97] J.M. Norton, J. Alzerreca, Y. Suwa, M.G. Klotz, Diversity of ammonia monooxygenase operon in autotrophic ammonia-oxidizing bacteria, Arch. Microbiol. 177 (2002) 139–149.
- [98] N.G. Hommes, L.A. Sayavedra-Soto, D.J. Arp, Transcript analysis of multiple copies of *amo* (encoding ammonia monooxygenase) and *hao* (encoding hydroxylamine oxidoreductase) in *Nitrosomonas europaea*, J. Bacteriol. 183 (2001) 1096–1100.
- [99] L.A. Sayavedra-Soto, N.G. Hommes, J.J. Alzerreca, D.J. Arp, J.M. Norton, M.G. Klotz, Transcription of the amoC, amoA and amoB genes in Nitrosomonas europaea and Nitrosospira sp. NpAV, FEMS Microbiol. Lett. 167 (1998) 81–88.
- [100] A.F. El Sheikh, M.G. Klotz, Ammonia-dependent differential regulation of the gene cluster that encodes ammonia monooxygenase in *Nitrosococcus oceani* ATCC 19707, Environ. Microbiol. 10 (2008) 3026–3035.
- [101] A.F. El Sheikh, A.T. Poret-Peterson, M.G. Klotz, Characterization of two new genes, amoR and amoD, in the amo operon of the marine ammonia oxidizer Nitrosococcus oceani ATCC 19707, Appl. Environ. Microbiol. 74 (2008) 312–318.
- [102] P.M. Berube, D.A. Stahl, The divergent AmoC<sub>3</sub> subunit of ammonia monooxygenase functions as part of a stress response system in *Nitrosomonas europaea*, J. Bacteriol. 194 (2012) 3448–3456.
- [103] M.G. Klotz, J.M. Norton, Multiple copies of ammonia monooxygenase (amo) operons have evolved under biased AT/GC mutational pressure in ammonia-oxidizing autotrophic bacteria, FEMS Microbiol. Lett. 168 (1998) 303–311.
- [104] R. Balasubramanian, S.M. Smith, S. Rawat, L.A. Yatsunyk, T.L. Stemmler, A.C. Rosenzweig, Oxidation of methane by a biological dicopper centre, Nature 465 (2010) 115–119.
- [105] M. Martinho, D.-W. Choi, A.A. DiSpirito, W.E. Antholine, J.D. Semrau, E. Munck, Mössbauer studies of the membrane-associated methane monooxygenase from *Methylococcus capsulatus* Bath: evidence for a diiron center, J. Am. Chem. Soc. 129 (2007) 15783–15785.

- [106] R. Bartossek, A. Spang, G. Weidler, A. Lanzen, C. Schleper, Metagenomic analysis of ammonia oxidizing Archaea affiliated with the soil group, Front. Microbiol. 3 (2012) 208.
- [107] M. Whittaker, D. Bergmann, D. Arciero, A.B. Hooper, Electron transfer during the oxidation of ammonia by the chemolithotrophic bacterium *Nitrosomonas europaea*, Biochim. Biophys. Acta 1459 (2000) 346–355.
- [108] N. Igarashi, H. Moriyama, T. Fujiwara, Y. Fukumori, N. Tanaka, The 2.8 Å structure of hydroxylamine oxidoreductase from a nitrifying chemolithotrophic bacterium, *Nitrosomonas europaea*, Nat, Struct, Biol. 4 (1997) 276–284.
- [109] J. Kostera, M. Youngblut, J. Slosarczyk, A. Pacheco, Kinetic and product distribution analysis of NO• reductase activity in *Nitrosomonas europaea* hydroxylamine oxidoreductase, J. Biol. Inorg. Chem. 13 (2008) 1073–1083.
- [110] A. Pacheco, J. McGarry, J. Kostera, A. Corona, Techniques for investigating hydroxylamine disproportionation by hydroxylamine oxidoreductases, Meth. Enzymol. 486 (2011) 447–463.
- [111] D. Arciero, C. Balny, A.B. Hooper, Spectroscopic and rapid kinetic studies of reduction of cytochrome c554 by hydroxylamine reductase from *Nitrosomonas europaea*, J. Biol. Chem. 269 (1991) 11878–11886.
- [112] T. Iverson, D.M. Arciero, B.T. Hsu, M.S.P. Logan, A.B. Hooper, D.C. Rees, Heme packing motifs revealed by the crystal structure of the tetra-heme cytochrome c554 from Nitrosomonas europaea, Nat. Struct. Biol. 5 (1998) 1005–1012.
- [113] T. Iverson, D.M. Arciero, A.B. Hooper, D.C. Rees, High-resolution structures of the oxidized and reduced states of cytochrome c554 from Nitrosomonas europaea, J. Biol. Inorg. Chem. 6 (2001) 390–397.
- [114] M.G. Klotz, Y.L. Stein, Genomics of ammonia-oxidizing bacteria and insights to their evolution, in: B.B. Ward, D.J. Arp, M.G. Klotz (Eds.), Nitrification, ASM Press, Washington, D.C., 2011, pp. 57–93.
- [115] D.J. Bergmann, D. Arciero, A.B. Hooper, Organization of the *hao* gene cluster of *Nitrosomonas europaea*: genes for two tetraheme *c* cytochromes, J. Bacteriol. 176 (1994) 3148–3153.
- [116] L.A. Sayavedra-Soto, N.G. Hommes, S.A. Russel, D.J. Arp, Induction of ammonia monooxygenase and hydroxylamine reductase mRNAs by ammonium in *Nitrosomonas europaea*, Mol. Microbiol. 20 (1996) 541–548.
- [117] M.A. Campbell, L.Y. Stein, M.G. Klotz, Ammonium- and hydroxylamine-induced transcriptomes differentiate between nitrogen- and energy-mediated gene expression in *Nitrosococcus oceani* ATCC 19707, Front. Microbiol. in review.
- [118] J.E. Graham, N.B. Wantland, M. Campbell, M.G. Klotz, Characterizing bacterial gene expression in nitrogen cycle metabolism with RT-qPCR, Meth. Enzymol. 496 (2011) 345–372.
- [119] A.T. Poret-Peterson, J.E. Graham, J. Gulledge, M.G. Klotz, Transcription of nitrification genes by the methane-oxidizing bacterium, *Methylococcus capsulatus* strain Bath, ISME J. 2 (2008) 1213–1220.
- [120] R.A. Hirota, A. Kuroda, T. Ikeda, N. Takiguchi, H. Ohtake, J. Kato, Transcriptional analysis of the multicopy *hao* gene coding for hydroxylamine oxidoreductase in *Nitrosomonas* sp. Strain ENI-11, Biosci. Biotechnol. Biochem. 70 (2006) 1875–1881.
- [121] M.S.M. Jetten, M. Strous, K.T. van de Pas-Schoonen, J. Schalk, U.G.J.M. van Dongen, A.A. van de Graaf, S. Logemann, G. Muyzer, M.C.M. van Loosdrecht, J.G. Kuenen, The anaerobic oxidation of ammonium, FEMS Microbiol. Rev. 22 (1998) 421–437.
- [122] J.S. Schalk, S. de Vries, J.G. Kuenen, M.S.M. Jetten, Involvement of a novel hydroxylamine oxidoreductase in anaerobic ammonium oxidation, Biochemistry 39 (2000) 5405–5412.
- [123] M. Strous, E. Pelletier, S. Mangenot, T. Rattei, A. Lehner, M.W. Taylor, M. Horn, H. Daims, D. Bartol-Mavel, P. Wincker, V.R. Barbe, N. Fonknechten, D. Vallenet, B.A. Segurens, C. Schenowitz-Truong, C. Medigue, A. Collingro, B. Snel, B.E. Dutilh, H.J.M. Op den Camp, C. van der Drift, I. Cirpus, K.T. van de Pas-Schoonen, H.R. Harhangi, L. van Niftrik, M. Schmid, J. Keltjens, J. van de Vossenberg, B. Kartal, H. Meier, D. Frishman, M.A. Huynen, H.-W. Mewes, J. Weissenbach, M.S.M. Jetten, M. Wagner, D. Le Paslier, Deciphering the evolution and metabolism of an anammox bacterium from a community genome, Nature 440 (2006) 790–794.
- [124] J.G. Kuenen, Anammox bacteria: from discovery to application, Nat. Rev. Microbiol. 6 (2008) 320–326.
- [125] M.S.M. Jetten, L. van Niftrik, M. Strous, B. Kartal, J.T. Keltjens, H.J.M. Op den Camp, Biochemistry and molecular biology of anammox bacteria, Crit. Rev. Biochem. Mol. Biol. 44 (2009) 65–84.
- [126] M.C. Schmid, A.B. Hooper, M.G. Klotz, A. Pommerening-Roeser, H.J.M. op den Camp, M.S.M. Jetten, Environmental detection of the octaheme cyctochrome *c* hydroxylamine/hydrazine oxidoreductase genes of aerobic and anaerobic ammonium-oxidizing bacteria, Environ. Microbiol. 10 (2008) 3140–3149.
- [127] E. Bock, H. Sundermeyer-Klinger, E. Stackebrandt, New facultative lithoautotrophic nitrite oxidizing bacteria, Arch. Microbiol. 136 (1983) 281–284.
- [128] E. Bock, H.-P. Koops, H. Harms, B. Ahlers, The biochemistry of nitrifying organisms, in: J.M. Shively, L.L. Barton (Eds.), Variations in Autotrophic Life, Academic Press Limited, San Diego, 1991, pp. 171–200.
- [129] S. Ehrich, D. Behrens, E. Lebedeva, W. Ludwig, E. Bock, A new obligately chemolithoautotrophic, nitrite-oxidizing bacterium, *Nitrospira moscoviensis* sp. nov. and its phylogenetic relationship, Arch. Microbiol. 164 (1995) 16–23.
- [130] E.V. Lebedeva, M. Alawi, C. Fiencke, B. Namsaraev, E. Bock, E. Spieck, Moderately thermophilic nitrifying bacteria from a hot spring of the Baikal rift zone, FEMS Microbiol. Ecol. 54 (2005) 297–306.
- [131] E. Spieck, A. Lipski, Cultivation, growth physiology, and chemotaxonomy of nitrite-oxidizing bacteria, Meth. Enzymol. 486 (2011) 109–130.
- [132] S.R. Starkenburg, P.S.G. Chain, L.A. Sayavedra-Soto, L. Hauser, M.L. Land, F.W. Larimer, S.A. Malfatti, M.G. Klotz, P.J. Bottomley, D.J. Arp, W.J. Hickey, Genome sequence of the chemolithoautotrophic nitrite-oxidizing bacterium *Nitrobacter winogradskyi* Nb-255, Appl. Environ. Microbiol. 72 (2006) 2050–2063.
- [133] S.R. Starkenburg, F.W. Larimer, L.Y. Stein, M.G. Klotz, P.S.G. Chain, L.A. Sayavedra-Soto, A.T. Poret-Peterson, M.E. Gentry, D.J. Arp, B. Ward, P.J. Bottomley, Complete genome

sequence of *Nitrobacter hamburgensis* X14 and comparative genomic analysis of species within the genus *Nitrobacter*, Appl. Environ. Microbiol. 74 (2008) 2852–2863.

- [134] N.M. de Almeida, W.J. Maalcke, J.T. Keltjens, M.S.M. Jetten, B. Kartal, Proteins and protein complexes involved in the biochemical reactions of anaerobic ammoniumoxidizing bacteria, Biochem. Soc. Trans. 39 (2011) 303–308.
- [135] A. Freitag, M. Rudert, E. Bock, Growth of Nitrobacter by dissimilatory nitrate reduction, FEMS Microbiol. Lett. 48 (1987) 105–109.
- [136] K. Kirstein, E. Bock, Close genetic relationship between Nitrobacter hamburgensis nitrite oxidoreductase and Escherichia coli nitrate reductases, Arch. Microbiol. 160 (1993) 447–453.
- [137] B. Kartal, M.M.M. Kuypers, G. Lavik, H.J.M. Op den Camp, M.S.M. Jetten, M. Strous, Anammox bacteria disguised as denitrifiers: nitrate reduction to dinitrogen gas via nitrite and ammonium, Environ. Microbiol. 9 (2007) 635–642.
- [138] M.M. Pereira, P.N. Refojo, G.O. Hreggvidsson, S. Hjorleifsdottir, M. Teixeira, The alternative complex III from *Rhodothermus marinus*—a prototype of a new family of quinol:electron acceptor oxidoreductases, FEBS Lett. 581 (2007) 4831–4835.
- [139] R. Gross, R. Pisa, M. Sänger, C.R.D. Lancaster, J. Simon, Characterization of the menaquinone reduction site in the diheme cytochrome *b* membrane anchor of *Wolinella succinogenes* NiFe-hydrogenase, J. Biol. Chem. 279 (2004) 274–281.
- [140] A. Kröger, S. Biel, J. Simon, R. Gross, G. Unden, C.R.D. Lancaster, Fumarate respiration of *Wolinella succinogenes*: enzymology, energetics and coupling mechanism, Biochim. Biophys. Acta 1553 (2002) 23–38.
- [141] L. Meek, D.J. Arp, The hydrogenase cytochrome *b* heme ligands of *Azotobacter vinelandii* are required for full H<sub>2</sub> oxidation capability, J. Bacteriol. 182 (2000) 3429–3436.
- [142] P.L. Tavormina, V.J. Orphan, M.G. Kalyuzhnaya, M.S.M. Jetten, M.G. Klotz, A novel family of functional operons encoding methane/ammonia monooxygenase-related proteins in gammaproteobacterial methanotrophs, Environ. Microbiol. Rep. 3 (2011) 91–100.
- [143] R. van Lis, A.-L. Ducluzeau, W. Nitschke, B. Schoepp-Cothenet, The nitrogen cycle in the Archaean: an intricate interplay of enzymatic and abiotic reactions, in: J.W.B. Moir (Ed.), Nitrogen Cycling in Bacteria, Molecular Analysis, Caister Academic Press, Norfolk, UK, 2011, pp. 1–21.
- [144] J.A. Imlay, Cellular defenses against superoxide and hydrogen peroxide, Annu. Rev. Biochem. 77 (2008) 755–776.
- [145] W. Martin, J. Baross, D. Kelley, M.J. Russell, Hydrothermal vents and the origin of life, Nat. Rev. Microbiol. 6 (2008) 805–814.
- [146] R. Gross, J. Simon, C.R.D. Lancaster, A. Kröger, Identification of histidine residues in *Wolinella succinogenes* hydrogenase that are essential for menaquinone reduction by H<sub>2</sub>, Mol. Microbiol. 30 (1998) 639–646.
- [147] M. Kern, J. Simon, Characterization of the NapGH quinol dehydrogenase complex involved in *Wolinella succinogenes* nitrate respiration, Mol. Microbiol. 69 (2008) 1137–1152.
- [148] T.H.C. Brondijk, A. Nilavongse, N. Filenko, D.J. Richardson, J.A. Cole, NapGH components of the periplasmic nitrate reductase of *Escherichia coli* K-12: location, topology and physiological roles in quinol oxidation and redox balancing, Biochem. J. 379 (2004) 47–55.
- [149] H. Hussain, J. Grove, L. Griffiths, S. Busby, J. Cole, A seven-gene operon essential for formate-dependent nitrite reduction to ammonia by enteric bacteria, Mol. Microbiol. 12 (1994) 153–163.
- [150] O. Klimmek, W. Dietrich, F. Dancea, Y.-J. Lin, S. Pfeiffer, F. Löhr, H. Rüterjans, R. Gross, J. Simon, A. Kröger, Sulfur respiration, in: D. Zannoni (Ed.), Respiration in Archaea and Bacteria. Vol. 2: Diversity of Prokaryotic Respiratory Systems, Springer, Dordrecht, 2004, pp. 217–232.
- [151] P.N. Refojo, F.L. Sousa, M. Teixeira, M.M. Pereira, The alternative complex III: a different architecture using known building blocks, Biochim. Biophys. Acta 1797 (2010) 1869–1876.
- [152] S.S. Venceslau, R.R. Lino, I.A.C. Pereira, The Qrc membrane complex, related to the alternative complex III, is a menaquinone reductase involved in sulfate respiration, J. Biol. Chem. 285 (2010) 22774–22783.
- [153] J. Simon, R. Pisa, T. Stein, R. Eichler, O. Klimmek, R. Gross, The tetraheme cytochrome *c* NrfH is required to anchor the cytochrome *c* nitrite reductase (NrfA) in the membrane of *Wolinella succinogenes*, Eur. J. Biochem. 268 (2001) 5776–5782.
- [154] S. Gon, M.T. Giudici-Orticoni, V. Méjean, C. lobbi-Nivol, Electron transfer and binding of the *c*-type cytochrome TorC to the trimethylamine N-oxide reductase in *Escherichia coli*, J. Biol. Chem. 276 (2001) 11545–11551.
- [155] M.L. Cartron, M.D. Roldán, S.J. Ferguson, B.C. Berks, D.J. Richardson, Identification of two domains and distal histidine ligands to the four hemes in the bacterial *c*-type cytochrome NapC: the prototyp connector between quinol/quinone and periplasmic oxido-reductases, Biochem. J. 368 (2002) 425–432.
- [156] M. Kern, O. Einsle, J. Simon, Variants of the tetrahaem cytochrome c quinol dehydrogenase NrfH characterize the menaquinol-binding site, the haem c-binding motifs and the transmembrane segment, Biochem. J. 414 (2008) 73–79.
- [157] R. Gross, R. Eichler, J. Simon, Site-directed modifications indicate differences in axial haem c iron ligation between the related NrfH and NapC families of multihaem c-type cytochromes, Biochem. J. 390 (2005) 689–693.
- [158] S.J. Marritt, T.G. Lowe, J. Bye, D.G.G. McMillan, L. Shi, J. Fredrickson, J. Zachara, D.J. Richardson, M.R. Cheesman, L.J.C. Jeuken, J.N. Butt, A functional description of CymA, an electron-transfer hub supporting anaerobic respiratory flexibility in *Shewanella*, Biochem. J. 444 (2012) 465–474.
- [159] D.G.G. McMillan, S.J. Marritt, J.N. Butt, L.J.C. Jeuken, Menaquinone-7 is specific cofactor in tetraheme quinol dehydrogenase CymA, J. Biol. Chem. 287 (2012) 14215–14225.
- [160] M.L. Rodrigues, K.A. Scott, M.S. Sansom, I.A. Pereira, M. Archer, Quinol oxidation by c-type cytochromes: structural characterization of the menaquinol binding site of NrfHA, J. Mol. Biol. 381 (2008) 341–350.

- [161] J. Simon, R. Eichler, R. Pisa, S. Biel, R. Gross, Modification of heme c binding motifs in the small subunit (NrfH) of the Wolinella succinogenes cytochrome c nitrite reductase complex, FEBS Lett. 522 (2002) 83–87.
- [162] S. Shirodkar, S. Reed, M. Romine, D. Saffarini, The octahaem SirA catalyses dissimilatory sulfite reduction in *Shewanella oneidensis* MR-1, Environ. Microbiol. 13 (2011) 108–115.
- [163] D. Sorokin, S. Lücker, D. Vejmelkova, N. Kostrikina, R. Kleerebezem, I.C. Rijpstra, J. Sinninghe Damste, D. Le Paslier, G. Muyzer, M. Wagner, M. van Loosdrecht, H. Daims, Nitrification expanded: discovery, physiology, and genomics of a nitrite-oxidizing bacterium from the phylum *Chloroflexi*, ISME J. (in press), http://dx.doi.org/10.1038/ismej.2012.70
- [164] M. Wikström, Active site intermediates in the reduction of O<sub>2</sub> by cytochrome oxidase, and their derivatives, Biochim. Biophys. Acta 1817 (2012) 468–475.
- [165] H. Han, J. Hemp, L.A. Pace, H. Ouyang, K. Ganesan, J.H. Roh, F. Daldal, S.R. Blanke, R.B. Gennis, Adaptation of aerobic respiration to low O<sub>2</sub> environments, Proc. Natl. Acad. Sci. U. S. A. 108 (2011) 14109–14114.
- [166] F.L. Sousa, R.J. Alves, M.A. Ribeiro, J.B. Pereira-Leal, M. Teixeira, M.M. Pereira, The superfamily of heme-copper oxygen reductases: types and evolutionary considerations, Biochim. Biophys. Acta 1817 (2012) 629–637.
- [167] A.J. Holmes, A. Costello, M.E. Lidstrom, J.C. Murrell, Evidence that particulate methane monooxygenase and ammonia monooxygenase may be evolutionarily related, FEMS Microbiol. Lett. 132 (1995) 203–208.
- [168] A. Pol, K. Heijmans, H.R. Harhangi, D. Tedesco, M.S.M. Jetten, H.J.M. Op den Camp, Methanotrophy below pH 1 by a new *Verrucomicrobia* species, Nature 450 (2007) 874–878.
- [169] T. Suzuki, T. Nakamura, H. Fuse, Isolation of two novel marine ethyleneassimilating bacteria, *Haliea* species ETY-M and ETY-NAG, containing particulate methane monooxygenase-like genes, Microb. Environ. 27 (2012) 54–60.
- [170] L.A. Sayavedra-Soto, N. Hamamura, C.-W. Liu, J.A. Kimbrel, J.H. Chang, D.J. Arp, The membrane-associated monooxygenase in the butane-oxidizing Gram-positive bacterium *Nocardioides* sp. strain CF8 is a novel member of the AMO/PMO family, Environ. Microbiol. Rep. 3 (2011) 1758–2229.
- [171] N.V. Coleman, N.B. Le, M.A. Ly, H.E. Ogawa, V. McCarl, N.L. Wilson, A.J. Holmes, Hydrocarbon monooxygenase in *Mycobacterium*: recombinant expression of a member of the ammonia monooxygenase superfamily, ISME J. 6 (2012) 171–182.
- [172] D. Canfield, M. Rosing, C. Bjerrum, Early anaerobic metabolisms, Philos. Trans. R. Soc. London, Ser. B 361 (2006) 1819–1836.
- [173] D.E. Canfield, A new model for Proterozoic ocean chemistry, Nature 396 (1998) 450–453.
- [174] P.G. Falkowski, Evolution of the nitrogen cycle and its influence on the biological sequestration of CO<sub>2</sub> in the ocean, Nature 387 (1997) 272–275.
- [175] L.R. Kump, The rise of atmospheric oxygen, Nature 451 (1993) 277–288.
- [176] S.E. Vlaeminck, A.G. Hay, L. Maignien, W. Verstraete, In quest of the nitrogen oxidizing prokaryotes of the early earth, Environ. Microbiol. 13 (2011) 283–295.
   [177] J.F. Kasting, Earth's early atmosphere, Science 259 (1993) 920–926.
- [177] J.F. Rasting, Early atmosphere, schene 259 (1995) 920–920.
   [178] A.-L. Ducluzeau, R. van Lis, S. Duval, B. Schoepp-Cothenet, M.J. Russell, W. Nitschke, Was nitric oxide the first deep electron sink? Trends Biochem. Sci. 34 (2009) 9–15.
- [179] LV. Godfrey, P.G. Falkowski, The cycling and redox state of nitrogen in the Archaean ocean, Nat. Geosci. 2 (2009) 725–729.
- [180] D.E. Canfield, A.N. Glazer, P.G. Falkowski, The evolution and future of Earth's nitrogen cycle, Science 330 (2010) 192–196.
- [181] A.D. Anbar, Elements and evolution, Science 322 (2008) 1481-1483.
- [182] S.W. Poulton, P.W. Fralick, D.E. Canfield, The transition to a sulphidic ocean ap-
- proximately 1.84 billion years ago, Nature 431 (2004) 173–177.
  [183] P. Lukat, M. Rudolf, P. Stach, A. Messerschmidt, P.M.H. Kroneck, J. Simon, O. Einsle, Binding and reduction of sulfite by cytochrome *c* nitrite reductase, Biochemistry 47 (2008) 2080–2086.

- [184] R.S. Hartshorne, M. Kern, B. Meyer, T.A. Clarke, M. Karas, D.J. Richardson, J. Simon, A dedicated haem lyase is required for the maturation of a novel bacterial cytochrome c with unconventional covalent haem binding, Mol. Microbiol. 64 (2007) 1049–1060.
- [185] S. Hartshorne, D.J. Richardson, J. Simon, Multiple haem lyase genes indicate substrate specificity in cytochrome c biogenesis, Biochem. Soc. Trans. 34 (2006) 146–149.
- [186] R. Pisa, T. Stein, R. Eichler, R. Gross, J. Simon, The nrfl gene is essential for the attachment of the active site haem group of Wolinella succinogenes cytochrome c nitrite reductase, Mol. Microbiol. 43 (2002) 763–770.
- [187] D.J. Eaves, J. Grove, W. Staudenmann, P. James, P.P. James, R.K. Poole, S.A. White, I. Griffiths, J.A. Cole, Involvement of products of the *nrfEFG* genes in the covalent attachment of haem c to a novel cysteine-lysine motif in the cytochrome c<sub>552</sub> nitrite reductase from *Escherichia coli*, Mol. Microbiol. 28 (1998) 205–216.
- [188] M. Kern, F. Eisel, J. Scheithauer, R.G. Kranz, J. Simon, Substrate specificity of three cytochrome *c* haem lyase isoenzymes from *Wolinella succinogenes*: unconventional haem *c* binding motifs are not sufficient for haem *c* attachment by Nrfl and CcsA1, Mol. Microbiol. 75 (2010) 122–137.
- [189] J.M. Stevens, D.A. Mavridou, R. Hamer, P. Kritsiligkou, A.D. Goddard, S.J. Ferguson, Cytochrome c biogenesis System I, FEBS J. 278 (2011) 4170–4178.
- [190] J. Simon, L. Hederstedt, Composition and function of cytochrome c biogenesis System II, FEBS J. 278 (2011) 4179–4188.
- [191] S. Bali, S.J. Ferguson, Assembly of respiratory proteins of the nitrogen cycle, in: J.W.B. Moir (Ed.), Nitrogen Cycling in Bacteria, Molecular Analysis, Caister Academic Press, Norfolk, UK, 2011, pp. 163–175.
- [192] S. Bali, A.D. Lawrence, S.A. Lobo, L.M. Saraiva, B.T. Golding, D.J. Palmer, M.J. Howard, S.J. Ferguson, M.J. Warren, Molecular hijacking of siroheme for the synthesis of heme and d<sub>1</sub> heme, Proc. Natl. Acad. Sci. U. S. A. 108 (2011) 18260–18265.
- [193] J. Hemp, H. Han, J.H. Roh, S. Kaplan, T.J. Martinez, R.B. Gennis, Comparative genomics and site-directed mutagenesis support the existence of only one input channel for protons in the C-family (*cbb*<sub>3</sub> oxidase) of heme-copper oxygen reductases, Biochemistry 46 (2007) 9963–9972.
- [194] J. Garvin, R. Buick, A.D. Anbar, G.L. Arnold, A.J. Kaufman, Isotopic evidence for an aerobic nitrogen cycle in the latest Archean, Science 323 (2009) 1045–1048.
- [195] C.D. Richter, J.W.A. Allen, C.W. Higham, A. Koppenhöfer, R.S. Zajicek, N.J. Watmough, S.J. Ferguson, Cytochrome cd<sub>1</sub>, reductive activation and kinetic analysis of a multifunctional respiratory enzyme, J. Biol. Chem. 27 (2002) 3093–3100.
- [196] D. Hira, H. Toh, C.T. Migita, H. Okubo, T. Nishiyama, M. Hattori, K. Furukawa, T. Fujii, Anammox organism KSU-1 expresses a NirK-type copper-containing nitrite reductase instead of a NirS-type with cytochrome cd<sub>1</sub>, FEBS Lett. 586 (2012) 1658–1663.
- [197] C.L. Dupont, A. Butcher, R.E. Valas, P.E. Bourne, G. Caetano-Anollés, History of biological metal utilization inferred through phylogenomic analysis of protein structures, Proc. Natl. Acad. Sci. U. S. A. 107 (2010) 10567–10572.
- [198] P.G. Falkowski, T. Fenchel, E.F. Delong, The microbial engines that drive Earth's biogeochemical cycles, Science 320 (2008) 1034–1039.
- [199] A.J. Thomson, G. Giannopoulos, J. Pretty, E.M. Baggs, D.J. Richardson, Biological sources and sinks of nitrous oxide and strategies to mitigate emissions, Philos. Trans. R. Soc. London, Ser. B 367 (2012) 1157–1168.
- [200] J. Rockstrom, W. Steffen, K. Noone, A. Persson, F.S. Chapin, E.F. Lambin, T.M. Lenton, M. Scheffer, C. Folke, H.J. Schellnhuber, B. Nykvist, C.A. de Wit, T. Hughes, S. van der Leeuw, H. Rodhe, S. Sorlin, P.K. Snyder, R. Costanza, U. Svedin, M. Falkenmark, L. Karlberg, R.W. Corell, V.J. Fabry, J. Hansen, B. Walker, D. Liverman, K. Richardson, P. Crutzen, J.A. Foley, A safe operating space for humanity, Nature 461 (2009) 472–475.
- [201] A. Marietou, D. Richardson, J. Cole, S. Mohan, Nitrate reduction by *Desulfovibrio desulfuricans*: a periplasmic nitrate reductase system that lacks NapB, but includes a unique tetraheme c-type cytochrome, NapM, FEMS Microbiol. Lett. 248 (2005) 217–225.